

This is a repository copy of *When does output feedback enlarge the capacity of the interference channel*?.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/122384/</u>

Version: Accepted Version

# Article:

Quintero, V., Perlaza, S.M., Esnaola, I. orcid.org/0000-0001-5597-1718 et al. (1 more author) (2018) When does output feedback enlarge the capacity of the interference channel? IEEE Transactions on Communications, 66 (2). ISSN 0090-6778

https://doi.org/10.1109/TCOMM.2017.2753252

© 2017 IEEE. Personal use of this material is permitted. Permission from IEEE must be obtained for all other users, including reprinting/ republishing this material for advertising or promotional purposes, creating new collective works for resale or redistribution to servers or lists, or reuse of any copyrighted components of this work in other works. Reproduced in accordance with the publisher's self-archiving policy.

# Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

# Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



# When Does Output Feedback Enlarge the Capacity of the Interference Channel?

Victor Quintero, Samir M. Perlaza, Iñaki Esnaola, Jean-Marie Gorce

Abstract—In this paper, the benefits of channel-output feedback in the Gaussian interference channel (G-IC) are studied under the effect of additive Gaussian noise. Using a linear deterministic (LD) model, the signal to noise ratios (SNRs) in the feedback links beyond which feedback plays a significant role in terms of increasing the individual rates or the sum-rate are approximated. The relevance of this work lies on the fact that it identifies the feedback SNRs for which in any G-IC one of the following statements is true: (a) feedback does not enlarge the capacity region; (b) feedback enlarges the capacity region and the sum-rate is greater than the largest sum-rate without feedback; and (c) feedback enlarges the capacity region but no significant improvement is observed in the sum-rate.

*Index Terms*—Interference Channel, Noisy Channel-Output Feedback, Capacity Region.

#### I. INTRODUCTION

The two-user Gaussian interference channel (G-IC) is the simplest channel model that captures the impairments brought by mutual interference into point-to-point communications subject to additive Gaussian noise. The interference channel (IC), in its most general form, was first proposed by Claude E. Shannon in [2]. The G-IC is a particular case that has been studied by several authors, see for instance [3]–[13] and references therein. However, despite this active research, the capacity region of the G-IC is characterized only in some special cases [4]. In general, the capacity region is not known exactly and only approximations to within a constant number of bits per channel-use per user are known [9].

On the other hand, channel-output feedback, which consists in letting a transmitter observe the channel-output at its intended receiver, was one of the first models for studying two-way point-to-point communications [14]. A G-IC with channel-output feedback is a model in which the backward direction (from receivers to transmitters) is exclusively used to let the transmitters observe the channel-output at the receivers with the goal of increasing the information rate or

Victor Quintero, Samir M. Perlaza and Jean-Marie Gorce are with the Laboratorie CITI (a joint laboratory between the Université de Lyon, INSA de Lyon, and INRIA). 6 Avenue des Arts, F-69621, Villeurbanne, France. ({victor.quintero-florez, samir.perlaza, jean-marie.gorce}@inria.fr).

Iñaki Esnaola is with the Department of Automatic Control and Systems Engineering, University of Sheffield, Mappin Street, Sheffield, S1 3JD, UK. (esnaola@sheffield.ac.uk).

Victor Quintero is also with Universidad del Cauca, Popayán, Colombia.

Samir M. Perlaza and Iñaki Esnaola are also with the Department of Electrical Engineering at Princeton University, Princeton, NJ, 08544, USA.

the reliability in the forward direction (from transmitters to receivers). Note that the backward direction may also be an IC since the point-to-point feedback links might be subject to mutual interference. There are several special cases of channeloutput feedback in the G-IC. First, the case in which the observation of the channel-output from the intended receiver is noiseless corresponds to perfect channel-output feedback (POF) [15]. Second, the case in which such observation is noisy corresponds to noisy channel-output feedback (NOF) [16], [17]. Third, the case in which such observation is a linear combination of the channel-outputs from both receivers subject to additive noise corresponds to wireless channeloutput feedback (WOF) [18]. The most general formulation is referred to as general channel-output feedback (GOF) [19]-[22]. Other types of feedback, including a channel-output processing, e.g., signal decoding, are known as rate-limited feedback (RLF) [23].

This work focuses in the case of G-IC with NOF (G-IC-NOF). One of the main motivations to focus on the G-IC-NOF stems from the recent findings regarding the impact of additive noise in the feedback links. In particular, in [16] and [17], it is shown that additive noise in the feedback links can dramatically change the number of generalized degrees of freedom (G-DoF) of the G-IC. In particular, one of the main benefits of feedback is that the number of G-DoF with perfect feedback increases monotonically with the interference to noise ratio (INR) in the very strong interference regime. However, in the presence of additive Gaussian noise in the feedback links, the number of G-DoF is bounded [16], [17].

#### A. Contributions

From the discussion above a relevant question arises: "When does channel-output feedback enlarge the capacity region of the G-IC?" This paper provides the answer when feedback links are impaired by noise and free of mutual interference, i.e., G-IC-NOF. The desired answer is of the form: "Implementing channel-output feedback in transmitter-receiver i enlarges the capacity region when the feedback SNR is greater than  $SNR_i^*$ , with  $i \in \{1, 2\}$  and fixed SNRs and INRs in the forward G-IC. Note that the description of the capacity region of the G-IC-NOF in [17] does not provide an answer to the question posed above. An answer in the desired form requires some calculations that, despite the conceptual simplicity of this analysis, are long and tedious. More specifically, the value  $SNR_i^*$  is obtained by comparing the capacity region of the linear deterministic IC (LD-IC) in [9] and the capacity region of the LD-IC with noisy channel-output feedback (LD-IC-NOF) in [17] to identify the feedback parameters that ensure

This research was supported in part by the European Commission under Marie Skłodowska-Curie Individual Fellowship No. 659316 (CYBERNETS); and the Administrative Department of Science, Technology, and Innovation of Colombia (Colciencias), fellowship No. 617-2013. Part of this work was presented at the 11th EAI International Conference on Cognitive Radio Oriented Wireless Networks (CROWNCOM), Grenoble, France, May 30-Jun 1 2016 [1].

strict inclusion of the former into the latter. After, using the fact that the capacity region of the LD-IC-NOF approximates the capacity region of the G-IC-NOF, an approximation of  $SNR_i^*$ is obtained. Solving this problem leads to a handful of equally relevant byproducts to determine whether or not implementing feedback in one of the transmitter-receiver pairs increases any of the individual rates or the sum-rate. That is, answers to the following questions: When does feedback in transmitterreceiver i allow achieving a rate  $R_1$ , such that for at least one  $R_2$ , all rate pairs  $(R'_1, R_2)$  achievable without feedback satisfy  $R_1 > R'_1$ ?; When does feedback in transmitter-receiver i allow achieving a rate  $R_2$ , such that for at least one  $R_1$ , all rate pairs  $(R_1, R'_2)$  achievable without feedback satisfy  $R_2 > R'_2$ ?; or When does feedback in transmitter-receiver i allow achieving a greater sum-rate than the maximum sumrate achievable without feedback?, with  $i \in \{1, 2\}$  and fixed SNRs and INRs in the forward G-IC.

The answers to the questions above provide consequential engineering insight about the benefits of feedback in the G-IC. For instance, all the cases in which feedback, even perfect channel-output feedback, is useless for increasing an individual rate or the sum-rate are identified. Similarly, this work provides guidelines for choosing in which of the pointto-point links feedback should be implemented for increasing either an individual rate or the sum-rate. Interestingly, in some cases, implementing feedback in only one of the transmitterreceiver pairs, despite the additive noise, turns out to be as beneficial as perfect channel-output feedback in both links.

#### B. Organization of the Paper

Section II introduces the G-IC and the linear deterministic IC (LD-IC). The capacity region of the G-IC is shown to be approximated by the capacity region of an LD-IC, with a particular choice of parameters. Section III presents the answers to the questions described above for the LD-IC. Section IV presents some LD-IC examples. Section V presents the implications of the conclusions obtained from the LD-IC (Section III) on the G-IC. The examples in Section IV are revisited in the context of the G-IC. The paper closes with the conclusions in Section VII.

#### **II. CHANNEL MODELS**

#### A. Gaussian Interference Channels

Consider the two-user G-IC-NOF depicted in Figure 1. Transmitter *i*, with  $i \in \{1, 2\}$ , communicates with receiver *i* subject to the interference produced by transmitter *j*, with  $j \in \{1, 2\} \setminus \{i\}$ . There are two independent and uniformly distributed messages,  $W_i \in W_i$ , with  $W_i = \{1, 2, \ldots, 2^{NR_i}\}$ , where *N* denotes the fixed blocklength in channel uses and  $R_i$  is the transmission rate in bits per channel use. At each block, transmitter *i* sends the codeword  $X_i = (X_{i,1}, X_{i,2}, \ldots, X_{i,N})^{\mathsf{T}} \in \mathcal{C}_i \subseteq \mathcal{X}_i^N$ , where  $\mathcal{X}_i$  and  $\mathcal{C}_i$  are respectively the channel-input alphabet and the codebook of transmitter *i*.

The channel coefficient from transmitter i to receiver i is denoted by  $\overrightarrow{h}_{ii}$ , the channel coefficient from transmitter j to receiver i is denoted by  $h_{ij}$ ; and the channel coefficient from



Fig. 1. Gaussian interference channel with noisy channel-output feedback at channel use n.

channel-output *i* to transmitter *i* is denoted by  $\overleftarrow{h}_{ii}$ . All channel coefficients are assumed to be non-negative real numbers. At a given channel use  $n \in \{1, 2, ..., N\}$ , the channel output at receiver *i* is denoted by  $\overrightarrow{Y}_{i,n}$ . During channel use *n*, the input-output relation of the channel model is given by

$$\overrightarrow{Y}_{i,n} = \overrightarrow{h}_{ii} X_{i,n} + h_{ij} X_{j,n} + \overrightarrow{Z}_{i,n}, \qquad (1)$$

where  $\vec{Z}_{i,n}$  is a real Gaussian random variable with zero mean and unit variance that represents the noise at the input of receiver *i*. Let d > 0 be the finite feedback delay measured in channel uses. At the end of channel use *n*, transmitter *i* observes  $Y_{i,n}$ , which consists of a scaled and noisy version of  $\vec{Y}_{i,n-d}$ . More specifically,

$$\overleftarrow{Y}_{i,n} = \begin{cases} \overleftarrow{Z}_{i,n} & \text{for } n \in \{1, 2, \dots, d\} \\ \overleftarrow{h}_{ii} \overrightarrow{Y}_{i,n-d} + \overleftarrow{Z}_{i,n}, & \text{for } n \in \{d+1, d+2, \dots, N\}, \end{cases}$$
(2)

where  $Z_{i,n}$  is a real Gaussian random variable with zero mean and unit variance that represents the noise in the feedback link of transmitter-receiver pair *i*. The random variables  $Z_{i,n}$ and  $Z_{i,n}$  are independent and identically distributed. In the following, without loss of generality, the feedback delay is assumed to be one channel use, i.e., d = 1. The encoder of transmitter *i* is defined by the set of deterministic functions  $\left\{f_i^{(1)}, f_i^{(2)}, \ldots, f_i^{(N)}\right\}$ , with  $f_i^{(1)} : \mathcal{W}_i \to \mathcal{X}_i$  and for  $n \in$  $\{2, 3, \ldots, N\}, f_i^{(n)} : \mathcal{W}_i \times \mathbb{R}^{n-1} \to \mathcal{X}_i$ , such that

$$X_{i,1} = f_i^{(1)}(W_i),$$
 (3a)

and for 
$$n \in \{2, 3, ..., N\}$$
.

$$X_{i,n} = f_i^{(n)} \left( W_i, \overleftarrow{Y}_{i,1}, \overleftarrow{Y}_{i,2}, \dots, \overleftarrow{Y}_{i,n-1} \right).$$
(3b)

The components of the input vector  $X_i$  are real numbers subject to an average power constraint:

$$\frac{1}{N}\sum_{n=1}^{N}\mathbb{E}\left(X_{i,n}^{2}\right) \leq 1,$$
(4)

where the expectation is taken over the joint distribution of the message indices  $W_1$ ,  $W_2$ , and the noise terms, i.e.,  $\vec{Z}_1$ ,  $\vec{Z}_2$ ,  $\vec{Z}_1$ , and  $\vec{Z}_2$ . The dependence of  $X_{i,n}$  on  $W_1$ ,  $W_2$ , and the previously observed noise realizations is due to the effect of feedback as shown in (2) and (3).

Hence, the decoder of receiver *i* is defined by the deterministic function  $\psi_i : \mathbb{R}_i^N \to \mathcal{W}_i$ . At the end of the communication, receiver *i* uses the vector  $\left(\overrightarrow{Y}_{i,1}, \overrightarrow{Y}_{i,2}, \ldots, \overrightarrow{Y}_{i,N}\right)^{\mathsf{T}}$  to obtain an estimate of the message index:

$$\widehat{W}_{i} = \psi_{i} \left( \overrightarrow{Y}_{i,1}, \overrightarrow{Y}_{i,2}, \dots, \overrightarrow{Y}_{i,N} \right),$$
(5)

where  $\widehat{W}_i$  is an estimate of the message index. The decoding error probability in the two-user G-IC-NOF, denoted by  $P_e(N)$ , is given by

$$P_e(N) = \max\left(\Pr\left(\widehat{W_1} \neq W_1\right), \Pr\left(\widehat{W_2} \neq W_2\right)\right). \quad (6)$$

The definition of an achievable rate pair  $(R_1, R_2) \in \mathbb{R}^2_+$  follows:

Definition 1 (Achievable Rate Pairs): A rate pair  $(R_1, R_2) \in \mathbb{R}^2_+$  is achievable if there exists at least one pair of codebooks in  $\mathcal{X}_1^N$  and in  $\mathcal{X}_2^N$  with codewords of length N, the corresponding encoding functions  $f_1^{(1)}, f_1^{(2)}, \ldots, f_1^{(N)}$  and  $f_2^{(1)}, f_2^{(2)}, \ldots, f_2^{(N)}$ , and the decoding functions  $\psi_1$  and  $\psi_2$ , such that the decoding error probability can be made arbitrarily small by letting the block-length N grow to infinity. The set of all achievable information rate pairs  $(R_1, R_2)$  is known as the information capacity region. The capacity region of a G-IC-NOF is described by six parameters:  $\overline{SNR}_i$ ,  $\overline{INR}_{ij}$  and  $\overline{SNR}_i$ , with  $i \in \{1, 2\}$  and  $j \in \{1, 2\} \setminus \{i\}$ , which are defined as follows:

$$\overrightarrow{\mathrm{SNR}}_i = \overrightarrow{h}_{ii}^2,\tag{7}$$

$$INR_{ij} = h_{ij}^2$$
, and (8)

$$\overleftarrow{\mathrm{SNR}}_{i} = \overleftarrow{h}_{ii}^{2} \left( \overrightarrow{h}_{ii}^{2} + 2 \overrightarrow{h}_{ii} h_{ij} + h_{ij}^{2} + 1 \right).$$
<sup>(9)</sup>

Given fixed parameters  $\overline{\text{SNR}}_1$ ,  $\overline{\text{SNR}}_2$ ,  $\text{INR}_{12}$ ,  $\text{INR}_{21}$ ,  $\overline{\text{SNR}}_1$ , and  $\overline{\text{SNR}}_2$ , the capacity region of the G-IC-NOF is approximated to within a constant number of bits by Theorem 4 in [17].

## B. Linear Deterministic Interference Channels

Consider the two-user LD-IC-NOF with parameters  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ ,  $n_{21}$ ,  $\vec{n}_{11}$  and  $\vec{n}_{22}$  depicted in Fig. 2. Parameter  $\vec{n}_{ii}$  represents the number of bit-pipes between transmitter *i* and receiver *i*; parameter  $n_{ij}$  represents the number of bitpipes between transmitter *j* and receiver *i*; and parameter  $\vec{n}_{ii}$  represents the number of bit-pipes between receiver *i* and transmitter *i* (feedback).

At transmitter *i*, the channel-input  $X_{i,n}$  during channel use *n*, with  $n \in \{1, 2, ..., N\}$ , is a *q*-dimensional binary vector  $X_{i,n} = \left(X_{i,n}^{(1)}, X_{i,n}^{(2)}, \dots, X_{i,n}^{(q)}\right)^{\mathsf{T}}$ , where

$$q = \max\left(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}\right), \qquad (10)$$

and N is the block-length. At receiver *i*, the channel-output  $\vec{Y}_{i,n}$  during channel use *n* is also a *q*-dimensional binary



Fig. 2. Two-user linear deterministic interference channel with noisy channeloutput feedback. The bit-pipe line number 1 represents the most significant bit.

vector  $\overrightarrow{\boldsymbol{Y}}_{i,n} = \left(\overrightarrow{\boldsymbol{Y}}_{i,n}^{(1)}, \overrightarrow{\boldsymbol{Y}}_{i,n}^{(2)}, \dots, \overrightarrow{\boldsymbol{Y}}_{i,n}^{(q)}\right)^{\mathsf{T}}$ . Let  $\boldsymbol{S}$  be a  $q \times q$  lower shift matrix of the form:

$$\boldsymbol{S} = \begin{bmatrix} 0 & 0 & 0 & \cdots & 0 \\ 1 & 0 & 0 & \cdots & 0 \\ 0 & 1 & 0 & \cdots & \vdots \\ \vdots & \ddots & \ddots & \ddots & 0 \\ 0 & \cdots & 0 & 1 & 0 \end{bmatrix}.$$
 (11)

The input-output relation during channel use n is given by

$$\overrightarrow{\boldsymbol{Y}}_{i,n} = \boldsymbol{S}^{q-\overrightarrow{n}_{ii}} \boldsymbol{X}_{i,n} + \boldsymbol{S}^{q-n_{ij}} \boldsymbol{X}_{j,n}, \qquad (12)$$

and the feedback signal  $\overline{Y}_{i,n}$  available at transmitter *i* at the end of channel use *n* satisfies

$$\overleftarrow{\mathbf{Y}}_{i,n} = \mathbf{S}^{\left(\max\left(\overrightarrow{n}_{ii}, n_{ij}\right) - \overleftarrow{n}_{ii}\right)^{+}} \overrightarrow{\mathbf{Y}}_{i,n-d}, \qquad (13)$$

where d is a finite delay, additions and multiplications are defined over the Galois Field of two elements GF(2), and  $(\cdot)^+$  is the positive part operator.

Without any loss of generality, the feedback delay is assumed to be equal to one channel use. Let  $W_i$  be the set of message indices of transmitter *i*. Transmitter *i* sends the message index  $W_i \in W_i$  by sending the codeword  $X_i = (X_{i,1}, X_{i,2}, \ldots, X_{i,N})$ , which is a binary  $q \times N$ matrix. The encoder of transmitter *i* can be modeled as a set of deterministic mappings  $\{f_i^{(1)}, f_i^{(2)}, \ldots, f_i^{(N)}\}$ , with  $f_i^{(1)} : W_i \to \{0, 1\}^q$  and for  $n \in \{2, 3, \ldots, N\}$ ,  $f_i^{(n)} :$  $W_i \times \{0, 1\}^{q \times (n-1)} \to \{0, 1\}^q$ , such that

$$\boldsymbol{X}_{i,1} = f_i^{(1)} \left( W_i \right) \tag{14a}$$

and for  $n \in \{2, 3, ..., N\}$ ,

$$\boldsymbol{X}_{i,n} = f_i^{(n)} \big( W_i, \overleftarrow{\boldsymbol{Y}}_{i,1}, \overleftarrow{\boldsymbol{Y}}_{i,2}, \dots, \overleftarrow{\boldsymbol{Y}}_{i,n-1} \big).$$
(14b)

The decoder of receiver *i* is defined by the deterministic function  $\psi_i : \{0, 1\}^{q \times N} \to \mathcal{W}_i$ . At the end of the communication, receiver *i* uses the sequence  $(\overrightarrow{Y}_{i,1}, \overrightarrow{Y}_{i,2}, \dots, \overrightarrow{Y}_{i,N})$  to obtain an estimate  $\widehat{W}_i$  of the message index  $W_i$ . The decoding error probability in the two-user LD-IC-NOF, denoted by  $P_e(N)$ , is given by (6).

A rate pair  $(R_1, R_2) \in \mathbb{R}^2_+$  is said to be achievable if it satisfies Definition 1. The set of all achievable information rate pairs  $(R_1, R_2)$  is known as the information capacity region and it is characterized by Theorem 1 in [17].

# C. Connections between Linear Deterministic and Gaussian Interference Channels

The capacity region of the G-IC-NOF with parameters  $\overrightarrow{SNR}_1$ ,  $\overrightarrow{SNR}_2$ ,  $\overrightarrow{INR}_{12}$ ,  $\overrightarrow{INR}_{21}$ ,  $\overrightarrow{SNR}_1$  and  $\overrightarrow{SNR}_2$  can be approximated by the capacity region of an LD-IC-NOF with parameters  $\overrightarrow{n}_{ii} = \lfloor \frac{1}{2} \log_2(\overrightarrow{SNR}_i) \rfloor$ ;  $n_{ij} = \lfloor \frac{1}{2} \log_2(\overrightarrow{INR}_{ij}) \rfloor$ ;  $\overleftarrow{n}_{ii} = \lfloor \frac{1}{2} \log_2(\overrightarrow{SNR}_i) \rfloor$ , with  $i \in \{1, 2\}$  and  $j \in \{1, 2\} \setminus \{i\}$ . For instance, in the case without feedback, the capacity region of any G-IC with parameters  $\overrightarrow{SNR}_1 > 1$ ,  $\overrightarrow{SNR}_2 > 1$ ,  $\overrightarrow{INR}_{12} > 1$  and  $\overrightarrow{INR}_{21} > 1$  is within 18.6 bits per channel use per user of the capacity of an LD-IC with parameters  $\overrightarrow{n}_{11} = \lfloor \frac{1}{2} \log_2(\overrightarrow{SNR}_1) \rfloor$ ,  $\overrightarrow{n}_{22} = \lfloor \frac{1}{2} \log_2(\overrightarrow{SNR}_2) \rfloor$ ,  $n_{12} = \lfloor \frac{1}{2} \log_2(\overrightarrow{INR}_{12}) \rfloor$ , and  $n_{21} = \lfloor \frac{1}{2} \log_2(\overrightarrow{INR}_{21}) \rfloor$  (Theorem 2 in [24]). More specifically, if the capacity region of the G-IC and the LD-IC without feedback are denoted by  $\mathcal{C}_G$  and  $\mathcal{C}_{LD}$ , respectively, the following holds:

$$C_{\rm LD} \subseteq C_{\rm G} + (5,5), \text{ and}$$
 (15a)

$$C_{\rm G} \subseteq C_{\rm LD} + (13.6, 13.6).$$
 (15b)

In a more general setting, for instance in the case with noisy channel-output feedback, the LD-IC is known to be a close approximation of the G-IC [17]. In Section V, this approximation is used to simplify the identification of the cases in which channel-output feedback, even subject to additive noise, enlarges the capacity region of the G-IC.

#### **III. MAIN RESULTS**

#### A. Preliminaries

Let  $\alpha_i \in \mathbb{Q}$ , with  $i \in \{1, 2\}$  and  $j \in \{1, 2\} \setminus \{i\}$  be defined as

$$\alpha_i = \frac{n_{ij}}{\overrightarrow{n}_{ii}}.$$
(16)

For each transmitter-receiver pair *i*, there exist five possible interference regimes (IRs), as suggested in [9]: the very weak IR (VWIR), i.e.,  $\alpha_i \leq \frac{1}{2}$ , the weak IR (WIR), i.e.,  $\frac{1}{2} < \alpha_i \leq \frac{2}{3}$ , the moderate IR (MIR), i.e.,  $\frac{2}{3} < \alpha_i < 1$ , the strong IR (SIR), i.e.,  $1 \leq \alpha_i \leq 2$  and the very strong IR (VSIR), i.e.,  $\alpha_i > 2$ . The scenarios in which the desired signal is stronger than the interference ( $\alpha_i < 1$ ), namely the VWIR, the WIR, and the MIR, are referred to as the low-interference regimes (LIRs). Conversely, the scenarios in which the desired signal is weaker than or equal to the interference ( $\alpha_i \geq 1$ ), namely the SIR and the VSIR, are referred to as the high-interference regimes (HIRs).

The main results of this paper are presented using a set of events (Boolean variables) that are determined by the parameters  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ , and  $n_{21}$ . Given a fixed tuple ( $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ ,  $n_{21}$ ), the events are defined below:

$$E_1: \qquad \alpha_1 < 1 \land \alpha_2 < 1, \qquad (17)$$

$$E_{2,i}: \qquad \alpha_i \leqslant \frac{1}{2} \land 1 \leqslant \alpha_j \leqslant 2, \tag{18}$$

$$E_{3,i}: \qquad \alpha_i \leqslant \frac{1}{2} \land \alpha_j > 2, \qquad (19)$$

$$E_{4,i}: \qquad \frac{1}{2} < \alpha_i \leqslant \frac{2}{3} \land \alpha_j \geqslant 1, \tag{20}$$

$$E_{5,i}: \qquad \frac{2}{3} < \alpha_i < 1 \land \alpha_j \ge 1, \tag{21}$$

$$E_{6,i}: \qquad \frac{1}{2} < \alpha_i \leqslant 1 \land \alpha_j > 1, \qquad (22)$$
  
$$E_{7,i}: \qquad \alpha_i \geqslant 1 \land \alpha_i \leqslant 1, \qquad (23)$$

$$E_{7,i}: \qquad \alpha_i \ge 1 \land \alpha_j \le 1, \qquad (23)$$
$$E_{8,i}: \qquad \overrightarrow{n}_{ii} > n_{ji}, \qquad (24)$$

$$E_9: \qquad \overrightarrow{n}_{11} + \overrightarrow{n}_{22} > n_{12} + n_{21}, \qquad (25)$$

$$E_{10,i}: \qquad \overrightarrow{n}_{ii} + \overrightarrow{n}_{jj} > n_{ij} + 2n_{ji}, \qquad (26)$$

$$E_{11,i}: \qquad \overrightarrow{n}_{ii} + \overrightarrow{n}_{jj} < n_{ij}. \tag{27}$$

In the following, in the case of  $E_{8,i}$ :  $\vec{n}_{ii} > n_{ji}$ , the notation  $\widetilde{E}_{8,i}$  indicates  $\vec{n}_{ii} < n_{ji}$ ; the notation  $\overline{E}_{8,i}$  indicates  $\vec{n}_{ii} \leq n_{ji}$  (logical complement); and the notation  $\check{E}_{8,i}$  indicates  $\vec{n}_{ii} \geq n_{ji}$ . In the case of  $E_1 : \alpha_1 < 1 \land \alpha_2 < 1$ , the notation  $\widetilde{E}_1$  indicates  $\alpha_1 > 1 \land \alpha_2 > 1$ ; and the notation  $\overline{E}_1$  indicates  $\alpha_1 \geq 1 \land \alpha_2 \geq 1$ . In the case of  $E_9 : \vec{n}_{11} + \vec{n}_{22} > n_{12} + n_{21}$ , the notation  $\overline{E}_9$  indicates  $\vec{n}_{11} + \vec{n}_{22} \leq n_{12} + n_{21}$ .

1

Combining the events (17)-(27), five main scenarios are identified:

$$S_{1,i}: (E_1 \wedge E_{8,i}) \vee (E_{2,i} \wedge E_{8,i}) \vee (E_{3,i} \wedge E_{8,i} \wedge E_9) \\ \vee (E_{4,i} \wedge E_{8,i} \wedge E_9) \vee (E_{5,i} \wedge E_{8,i} \wedge E_9), \qquad (28)$$
$$S_{2,i}: \left(E_{3,i} \wedge \widetilde{E}_{8,j} \wedge \overline{E}_9\right) \vee \left(E_{6,i} \wedge \widetilde{E}_{8,j} \wedge \overline{E}_9\right) \vee \left(\widetilde{E}_1 \wedge \widetilde{E}_{8,j}\right), \qquad (29)$$

$$S_{3,i}: (E_1 \wedge \overline{E}_{8,i}) \vee (E_{2,i} \wedge \overline{E}_{8,i}) \vee (E_{3,i} \wedge \check{E}_{8,j} \wedge \overline{E}_{8,i}) \\ \vee (E_{4,i} \wedge \check{E}_{8,j} \wedge \overline{E}_{8,i}) \vee (E_{5,i} \wedge \check{E}_{8,j} \wedge \overline{E}_{8,i}) \\ \vee (\overline{E}_1 \wedge \check{E}_{8,j}) \vee (E_{7,i}),$$

$$(30)$$

$$S_4 : E_1 \wedge E_{8,1} \wedge E_{8,2} \wedge E_{10,1} \wedge E_{10,2}, \tag{31}$$

$$S_5 : E_1 \wedge E_{11,1} \wedge E_{11,2}. \tag{32}$$

For all  $i \in \{1, 2\}$ , the events  $S_{1,i}, S_{2,i}, S_{3,i}, S_4$  and  $S_5$  exhibit the properties stated by the following corollaries.

Corollary 1: For all  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$ , given a fixed  $i \in \{1, 2\}$ , only one of the events  $S_{1,i}$ ,  $S_{2,i}$  and  $S_{3,i}$  holds true.

Corollary 2: For all  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$ , when one of the events  $S_4$  or  $S_5$  holds true, then the other necessarily holds false.

Note that Corollary 2 does not exclude the case in which both  $S_4$  and  $S_5$  simultaneously hold false.

Corollary 3: For all  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$ , when  $S_4$  holds true, then both  $S_{1,1}$  and  $S_{1,2}$  hold true; and when  $S_5$  holds true, then both  $S_{2,1}$  and  $S_{2,2}$  hold true.

#### **B.** Rate Improvement Metrics

Given a fixed tuple  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21})$ , let  $\mathcal{C}(\overleftarrow{n}_{11}, \overleftarrow{n}_{22})$ be the capacity region of an LD-IC with noisy channeloutput feedback with parameters  $\overleftarrow{n}_{11}$  and  $\overleftarrow{n}_{22}$ . The maximum improvement of the individual rates  $R_1$  and  $R_2$ , denoted by  $\Delta_1(\overleftarrow{n}_{11}, \overleftarrow{n}_{22})$  and  $\Delta_2(\overleftarrow{n}_{11}, \overleftarrow{n}_{22})$ , due to the effect of channel-output feedback with respect to the case without feedback is:

$$\Delta_{1}(\overleftarrow{n}_{11},\overleftarrow{n}_{22}) = \max_{0 < R_{2} < R_{2}^{*}} \left\{ \sup \left\{ R_{1} : (R_{1}, R_{2}) \in \mathcal{C}(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) \right\} - \sup \left\{ R_{1}^{\dagger} : (R_{1}^{\dagger}, R_{2}) \in \mathcal{C}(0, 0) \right\} \right\} \text{ and} \quad (33)$$

$$\Delta_{2}(\overleftarrow{n}_{11},\overleftarrow{n}_{22}) = \max_{0 < R_{1} < R_{1}^{*}} \left\{ \sup \left\{ R_{2} : (R_{1}, R_{2}) \in \mathcal{C}(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) \right\} - \sup \left\{ R_{2}^{\dagger} : (R_{1}, R_{2}^{\dagger}) \in \mathcal{C}(0, 0) \right\} \right\}, \quad (34)$$

with

$$R_1^* = \sup \{ r_1 : (r_1, r_2) \in \mathcal{C}(0, 0) \}$$
 and (35)

$$R_2^* = \sup \left\{ r_2 : (r_1, r_2) \in \mathcal{C}(0, 0) \right\}.$$
 (36)

Note that for a fixed  $i \in \{1, 2\}, \Delta_i(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) > 0$  if and only if it is possible to achieve a rate pair  $(R_1, R_2) \in \mathbb{R}^2_+$ with channel-output feedback such that  $R_i$  is greater than the maximum rate achievable by transmitter-receiver i without feedback when the rate of transmitter-receiver pair j is fixed at  $R_i$ . In the following, given fixed parameters  $\overleftarrow{n}_{11}$  and  $\overleftarrow{n}_{22}$ , the statement "the rate  $R_i$  is improved by using feedback" is used to indicate that  $\Delta_i(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) > 0$ .

Alternatively, the maximum improvement of the sum-rate  $\Sigma(\overline{n}_{11}, \overline{n}_{22})$  with respect to the case without feedback is:

$$\Sigma(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) = \sup \left\{ R_1 + R_2 : (R_1, R_2) \in \mathcal{C}(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) \right\}$$
$$- \sup \left\{ R_1^{\dagger} + R_2^{\dagger} : (R_1^{\dagger}, R_2^{\dagger}) \in \mathcal{C}(0, 0) \right\}. (37)$$

Note that  $\Sigma(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) > 0$  if and only if there exists a rate pair with feedback whose sum is greater than the maximum sum-rate achievable without feedback. In the following, given fixed parameters  $\overleftarrow{n}_{11}$  and  $\overleftarrow{n}_{22}$ , the statement "the sumrate is improved by using feedback" is used to imply that  $\Sigma(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) > 0.$ 

In the following, when feedback is exclusively used by transmitter-receiver pair *i*, i.e.,  $\overline{n}_{ii} > 0$  and  $\overline{n}_{jj} = 0$ , then the maximum improvement of the individual rate of transmitter-receiver k, with  $k \in \{1, 2\}$ , and the maximum improvement of the sum-rate are denoted by  $\Delta_k(\overline{n}_{ii})$  and  $\Sigma(\overline{n}_{ii})$ , respectively. Hence, this notation  $\Delta_k(\overline{n}_{ii})$  replaces either  $\Delta_k(\overline{n}_{11}, 0)$  or  $\Delta_k(0, \overline{n}_{22})$ , when i = 1 or i = 2, respectively. The same holds for the notation  $\Sigma(\overleftarrow{n}_{ii})$  that replaces  $\Sigma(\overleftarrow{n}_{11}, 0)$  or  $\Sigma(0, \overleftarrow{n}_{22})$ , when i = 1 or i = 2, respectively.

## C. Enlargement of the Capacity Region

Given fixed parameters  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}), i \in \{1, 2\},\$ and  $j \in \{1, 2\} \setminus \{i\}$ , the capacity region of a two-user LD-IC, when feedback is available only at transmitter-receiver pair i, i.e.,  $\overleftarrow{n}_{ii} > 0$  and  $\overleftarrow{n}_{jj} = 0$ , is denoted by  $\mathcal{C}(\overleftarrow{n}_{ii})$  instead of  $\mathcal{C}(\overleftarrow{n}_{11}, 0)$  or  $\mathcal{C}(0, \overleftarrow{n}_{22})$ , when i = 1 or i = 2, respectively. Following this notation, Theorem 1 identifies the exact values of  $\overline{n}_{ii}$  for which the strict inclusion  $\mathcal{C}(0,0) \subset \mathcal{C}(\overline{n}_{ii})$  holds for  $i \in \{1, 2\}$ .

Theorem 1: Let  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$  be a fixed tuple. Let also  $i \in \{1,2\}, j \in \{1,2\} \setminus \{i\}$  and  $\overleftarrow{n}_{ii}^* \in \mathbb{N}$  be fixed integers, with

$$\overleftarrow{n}_{ii}^* = \begin{cases} \max\left(n_{ji}, \left(\overrightarrow{n}_{ii} - n_{ij}\right)^+\right) & \text{if } S_{1,i} \text{ holds true} \\ \overrightarrow{n}_{jj} + \left(\overrightarrow{n}_{ii} - n_{ij}\right)^+ & \text{if } S_{2,i} \text{ holds true} \end{cases}.$$
(38)

Assume that  $S_{3,i}$  holds true. Then, for all  $\overleftarrow{n}_{ii} \in \mathbb{N}$ ,  $\mathcal{C}(0,0) = \mathcal{C}(\overleftarrow{n}_{ii})$ . Assume that either  $S_{1,i}$  holds true or  $S_{2,i}$ holds true. Then, for all  $\overleftarrow{n}_{ii} \leq \overleftarrow{n}_{ii}^*$ ,  $\mathcal{C}(0,0) = \mathcal{C}(\overleftarrow{n}_{ii})$  and for all  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^*$ ,  $\mathcal{C}(0,0) \subset \mathcal{C}(\overleftarrow{n}_{ii})$ . *Proof:* The proof of Theorem 1 is presented in Appendix

Α.

Theorem 1 shows that under event  $S_{3,i}$  in (30), implementing feedback in transmitter-receiver pair *i*, with any  $\overline{n}_{ii} > 0$ and  $\overline{n}_{ij} = 0$ , does not enlarge the capacity region. Note that when both  $E_{8,i}$  and  $E_{8,j}$  hold false, then both  $S_{1,i}$  and  $S_{2,i}$ hold false, which implies that  $S_{3,i}$  holds true (Corollary 1). The following remark is a consequence of this observation.

Remark 1: A necessary but not sufficient condition for enlarging the capacity region by using feedback in transmitterreceiver pair i is: there exists at least one transmitter able to send more information bits to receiver *i* than to receiver *j*, *i.e.*,  $\overrightarrow{n}_{ii} > n_{ji}$  (Event  $E_{8,i}$ ) or  $n_{ij} > \overrightarrow{n}_{jj}$  (Event  $E_{8,j}$ ).

Alternatively, under events  $S_{1,i}$  in (28) and  $S_{2,i}$  in (29), the capacity region can be enlarged when  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ij}^*$ . It is important to highlight that in the cases in which feedback enlarges the capacity region of the two-user LD-IC-NOF, that is, in events  $S_{1,1}$ ,  $S_{2,1}$ ,  $S_{1,2}$  or  $S_{2,2}$ , for all  $i \in \{1,2\}$  and  $j \in \{1, 2\} \setminus \{i\}$ , the following always holds true:

$$\overleftarrow{n}_{ii}^* > (\overrightarrow{n}_{ii} - n_{ij})^+.$$
(39)

Essentially, the inequality in (39) unveils a necessary but not sufficient condition to enlarge the capacity region using channel-output feedback. This condition is that for at least one  $i \in \{1, 2\}$ , with  $j \in \{1, 2\} \setminus \{i\}$ , transmitter i decodes a subset of the information bits sent by transmitter j at each channel use.

Another interesting observation is that the threshold  $\overleftarrow{n}_{ii}^*$ beyond which feedback is useful is different under event  $S_{1,i}$ in (28) and event  $S_{2,i}$  in (29). In general when  $S_{1,i}$  holds true, the enlargement of the capacity region is due to the fact that feedback allows using interference as side information [25]. Alternatively, when  $S_{2,i}$  in (29) holds true, the enlargement of the capacity region occurs as a consequence of the fact that some of the bits that cannot be transmitted directly from transmitter j to receiver j, can arrive to receiver j via an alternative path: transmitter j - receiver i - transmitter i receiver j. Both scenarios, interference as side information and alternative path, are extensively discussed in [15], [16], and [17].

#### D. Improvement of the Individual Rate $R_i$ by Using Feedback in Link i

Given fixed parameters  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21})$ , and  $i \in \{1, 2\}$ , implementing channel-output feedback in transmitterreceiver pair *i* increases the individual rate  $R_i$ , i.e.,  $\Delta_i(\overleftarrow{n}_{ii}) > 0$  for some values of  $\overleftarrow{n}_{ii}$ . Theorem 2 identifies the exact values of  $\overleftarrow{n}_{ii}$  for which  $\Delta_i(\overleftarrow{n}_{ii}) > 0$ .

Theorem 2: Let  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$  be a fixed tuple. Let also  $i \in \{1, 2\}, j \in \{1, 2\} \setminus \{i\}$  and  $\overleftarrow{n}_{ii}^{\dagger} \in \mathbb{N}$  be fixed integers, with

$$\overleftarrow{n}_{ii}^{\dagger} = \max\left(n_{ji}, \left(\overrightarrow{n}_{ii} - n_{ij}\right)^{+}\right).$$
(40)

Assume that either  $S_{2,i}$  holds true or  $S_{3,i}$  holds true. Then, for all  $\overleftarrow{n}_{ii} \in \mathbb{N}$ ,  $\Delta_i(\overleftarrow{n}_{ii}) = 0$ . Assume that  $S_{1,i}$  holds true. Then, when  $\overleftarrow{n}_{ii} \leq \overleftarrow{n}_{ii}^{\dagger}$ , it holds that  $\Delta_i(\overleftarrow{n}_{ii}) = 0$ ; and when  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^{\dagger}$ , it holds that  $\Delta_i(\overleftarrow{n}_{ii}) > 0$ .

*Proof:* The proof of Theorem 2 is presented in Appendix B.

Theorem 2 highlights that under events  $S_{2,i}$  in (29) and  $S_{3,i}$  in (30), the individual rate  $R_i$  cannot be improved by using feedback in transmitter-receiver pair *i*, i.e.,  $\Delta_i(\overleftarrow{n}_{ii}) = 0$ . Alternatively, under event  $S_{1,i}$  in (28), the individual rate  $R_i$  can be improved, i.e.,  $\Delta_i(\overleftarrow{n}_{ii}) > 0$ , whenever  $\overleftarrow{n}_{ii} > \max(n_{ji}, (\overrightarrow{n}_{ii} - n_{ij})^+)$ . Hence, given the definition of  $S_{1,i}$ , the following remark is relevant.

**Remark 2:** A necessary but not sufficient condition for  $\Delta_i(\overleftarrow{n}_{ii}) > 0$  is: the number of bit-pipes from transmitter *i* to receiver *i* is greater than the number of bit-pipes from transmitter *i* to receiver *j*, i.e.,  $\overrightarrow{n}_{ii} > n_{ji}$  (Event  $E_{8,i}$ )

## E. Improvement of the Individual Rate $R_j$ by Using Feedback in Link i

Given fixed parameters  $(\vec{n}_{11}, \vec{n}_{22}, n_{12}, n_{21}), i \in \{1, 2\}$ , and  $j \in \{1, 2\} \setminus \{i\}$ , implementing channel-output feedback in transmitter-receiver pair *i* increases the individual rate  $R_j$ , i.e.,  $\Delta_j(\overleftarrow{n}_{ii}) > 0$  for some values of  $\overleftarrow{n}_{ii}$ . Theorem 3 identifies the exact values of  $\overleftarrow{n}_{ii}$  for which  $\Delta_j(\overleftarrow{n}_{ii}) > 0$ .

Theorem 3: Let  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$  be a fixed tuple. Let also  $i \in \{1, 2\}, j \in \{1, 2\} \setminus \{i\}$  and  $\overleftarrow{n}_{ii}^* \in \mathbb{N}$ given in (38), be fixed integers. Assume that  $S_{3,i}$  holds true. Then, for all  $\overleftarrow{n}_{ii} \in \mathbb{N}, \Delta_j(\overleftarrow{n}_{ii}) = 0$ . Assume that either  $S_{1,i}$  holds true or  $S_{2,i}$  holds true. Then, when  $\overleftarrow{n}_{ii} \leq \overleftarrow{n}_{ii}^*$ , it holds that  $\Delta_j(\overleftarrow{n}_{ii}) = 0$ ; and when  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^*$ , it holds that  $\Delta_j(\overleftarrow{n}_{ii}) > 0$ .

*Proof:* The proof of Theorem 3 follows along the same lines of the proof of Theorem 2 in Appendix B.

Theorem 3 shows that under event  $S_{3,i}$  in (30), implementing feedback in transmitter-receiver pair *i* does not bring any improvement on the rate  $R_j$ . This is in line with the results of Theorem 1. In contrast, under events  $S_{1,i}$  in (28) and  $S_{2,i}$  in (29), the individual rate  $R_j$  can be improved, i.e.,  $\Delta_j(\overline{n}_{ii}) > 0$  for all  $\overline{n}_{ii} > \overline{n}_{ii}^*$ . From the definition of events  $S_{1,i}$  and  $S_{2,i}$ , the following remark holds:

**Remark** 3: A necessary but not sufficient condition for  $\Delta_j(\overleftarrow{n}_{ii}) > 0$  is: there exists at least one transmitter able

to send more information bits to receiver *i* than to receiver *j*, i.e.,  $\overrightarrow{n}_{ii} > n_{ji}$  (Event  $E_{8,i}$ ) or  $n_{ij} > \overrightarrow{n}_{jj}$  (Event  $\widetilde{E}_{8,j}$ ).

It is important to highlight that under event  $S_{1,i}$ , the threshold on  $\overline{n}_{ii}$  for increasing the individual rate  $R_i$ , i.e.,  $\overline{n}_{ii}^{\dagger}$ , and  $R_j$ , i.e.,  $\overline{n}_{ii}^{\ast}$ , are identical, see Theorem 2 and Theorem 3. This implies that in this case, the use of feedback in transmitter-receiver pair *i*, with  $\overline{n}_{ii} > \overline{n}_{ij}^{\dagger} = \overline{n}_{ii}^{\ast}$ , benefits both transmitter-receiver pairs, i.e.,  $\Delta_i(\overline{n}_{ii}) > 0$  and  $\Delta_j(\overline{n}_{ii}) > 0$ . Under event  $S_{2,i}$ , using feedback in transmitter-receiver pair *i*, with  $\overline{n}_{ii} > \overline{n}_{ii}^{\ast}$ , exclusively benefits transmitter-receiver pair *j*, i.e.,  $\Delta_i(\overline{n}_{ii}) = 0$  and  $\Delta_j(\overline{n}_{ii}) > 0$ .

#### F. Improvement of the Sum-Rate

transmitter-receiver pair *i*.

Given fixed parameters  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21})$ , and  $i \in \{1, 2\}$ , implementing channel-output feedback in transmitter-receiver pair *i* increases the sum-rate, i.e.,  $\Sigma(\overleftarrow{n}_{ii}) > 0$  for some values of  $\overleftarrow{n}_{ii}$ . Theorem 4 identifies the exact values of  $\overleftarrow{n}_{ii}$  for which  $\Sigma(\overleftarrow{n}_{ii}) > 0$ .

Theorem 4: Let  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$  be a fixed tuple. Let also  $i \in \{1, 2\}, j \in \{1, 2\} \setminus \{i\}$  and  $\overleftarrow{n}_{ii}^+ \in \mathbb{N}$  be fixed integers, with

$$\overleftarrow{n}_{ii}^{+} = \begin{cases} \max\left(n_{ji}, \left(\overrightarrow{n}_{ii} - n_{ij}\right)^{+}\right) & \text{if } S_4 \text{ holds true} \\ \overrightarrow{n}_{jj} + \left(\overrightarrow{n}_{ii} - n_{ij}\right)^{+} & \text{if } S_5 \text{ holds true} \\ \end{cases}$$

$$(41)$$

Assume that  $S_4$  holds false and  $S_5$  holds false. Then,  $\Sigma(\overleftarrow{n}_{ii}) = 0$  for all  $\overleftarrow{n}_{ii} \in \mathbb{N}$ . Assume that  $S_4$  holds true or  $S_5$ holds true. Then, when  $\overleftarrow{n}_{ii} \leq \overleftarrow{n}_{ii}^+$ , it holds that  $\Sigma(\overleftarrow{n}_{ii}) = 0$ ; and when  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^+$ , it holds that  $\Sigma(\overleftarrow{n}_{ii}) > 0$ .

*Proof:* The proof of Theorem 4 is presented in Appendix C. Theorem 4 introduces a necessary but not sufficient condition for improving the sum-rate by implementing feedback in

**Remark** 4: A necessary but not sufficient condition for observing  $\Sigma(\overleftarrow{n}_{ii}) > 0$  is to satisfy one of the following conditions: (a) both transmitter-receiver pairs are in LIR (Event  $E_1$ ); or (b) both transmitter-receiver pairs are in HIR (Event  $\overline{E}_1$ ).

Finally, it follows from Corollary 3 that when  $S_4$  or  $S_5$  holds true, with  $i \in \{1,2\}$  and  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^+$ , in addition to  $\Sigma(\overleftarrow{n}_{ii}) > 0$ , it also holds that  $\Delta_1(\overleftarrow{n}_{ii}) > 0$  and  $\Delta_2(\overleftarrow{n}_{ii}) > 0$ .

#### **IV. EXAMPLES**

*Example 1: Consider an LD-IC-NOF with parameters*  $\overrightarrow{n}_{11} = 7$ ,  $\overrightarrow{n}_{22} = 7$ ,  $n_{12} = 3$ , and  $n_{21} = 5$ .

In Example 1, both  $S_{1,1}$  and  $S_{1,2}$  hold true. Hence, from Theorem 1, when  $\overleftarrow{n}_{11} > 5$  or  $\overleftarrow{n}_{22} > 3$ , there always exists an enlargement of the capacity region. More specifically, it follows from Theorem 2 and Theorem 3 that using feedback in transmitter-receiver pair 1, with  $\overleftarrow{n}_{11} > 5$  or using feedback in transmitter-receiver pair 2, with  $\overleftarrow{n}_{22} > 3$ , both individual rates can be simultaneously improved, i.e.,  $\Delta_1(\overleftarrow{n}_{ii}) > 0$ and  $\Delta_2(\overleftarrow{n}_{ii}) > 0$  with i = 1 or i = 2, respectively. Alternatively, note that  $S_4$  holds true. Hence, it follows from



Fig. 3. Capacity regions C(0,0) (thick red line) and C(6,0) (thin blue line), with  $\vec{n}_{11} = 7$ ,  $\vec{n}_{22} = 7$ ,  $n_{12} = 3$ ,  $n_{21} = 5$ .

Theorem 4 that using feedback in transmitter-receiver pair 1, with  $\overline{m}_{11} > 5$  or using feedback in transmitter-receiver pair 2, with  $\overline{m}_{22} > 3$ , improves the sum-rate, i.e.,  $\Sigma(\overline{m}_{ii}) > 0$ with i = 1 or i = 2 respectively. These conclusions are observed in Figure 3, for the case  $\overline{m}_{11} = 6$  and  $\overline{m}_{22} = 0$ , where the capacity regions C(0,0) (thick red line) and C(6,0)(thin blue line) are plotted. Note that, when  $\overline{m}_{11} = 6$ , there always exist a rate pair  $(R'_1, R'_2) \in C(0,0)$  and a rate pair  $(R_1, R_2) \in C(6,0) \setminus C(0,0)$  such that  $R'_1 < R_1$  and  $R'_2 = R_2$ (Theorem 2). Simultaneously, there always exist a rate pair  $(R'_1, R'_2) \in C(0,0)$  and a rate pair  $(R_1, R_2) \in C(6,0) \setminus C(0,0)$ such that  $R'_2 < R_2$  and  $R'_1 = R_1$  (Theorem 3). Finally, note that for all rate pairs  $(R'_1, R'_2) \in C(0,0)$  there always exists a rate pair  $(R_1, R_2) \in C(6,0)$ , for which  $R_1 + R_2 > R'_1 + R'_2$ (Theorem 4).

*Example 2: Consider an LD-IC-NOF with parameters*  $\vec{n}_{11} = 7$ ,  $\vec{n}_{22} = 8$ ,  $n_{12} = 6$ , and  $n_{21} = 5$ .

In Example 2, the events  $S_{1,1}$  and  $S_{1,2}$  hold true; and the events  $S_4$  and  $S_5$  hold false. Hence, it follows from Theorem 4 that using feedback in either transmitter-receiver pair does not improve the sum-rate, i.e., for all  $i \in \{1,2\}$  and for all  $\overline{n}_{ii} > 0$ ,  $\Sigma(\overline{n}_{ii}) = 0$ . These conclusions are observed in Figure 4, for the case  $\overline{n}_{11} = 0$  and  $\overline{n}_{22} = 7$ , where the capacity regions C(0,0) (thick red line) and C(0,7) (thin blue line) are plotted. From Example 2, it becomes evident that when  $S_{1,1}$  and  $S_{1,2}$  hold true,  $S_4$  and  $S_5$  do not necessarily hold true. That is, the improvements on the individual rates, despite that they can be observed simultaneously, are not enough to improve the sum-rate beyond what is already achievable without feedback.

*Example 3: Consider an LD-IC-NOF with parameters*  $\overrightarrow{n}_{11} = 5$ ,  $\overrightarrow{n}_{22} = 1$ ,  $n_{12} = 3$ , and  $n_{21} = 4$ .

In Example 3, both  $S_{2,1}$  in (29) and  $S_{3,2}$  in (30) hold true. Hence, it follows from Theorem 1 that the capacity region can be enlarged by using feedback in transmitter-receiver pair 1 when  $\overleftarrow{n}_{11} > 3$ , whereas using feedback in transmitter-receiver pair 2 does not enlarge the capacity region. More specifically, it follows from Theorem 2 and Theorem 3 that using feedback in transmitter-receiver pair 1 does not improve the individual



Fig. 4. Capacity regions C(0,0) (thick red line) and C(0,7) (thin blue line), with  $\vec{n}_{11} = 7$ ,  $\vec{n}_{22} = 8$ ,  $n_{12} = 6$ ,  $n_{21} = 5$ .

rate  $R_1$  but  $R_2$ , i.e.,  $\Delta_1(\overleftarrow{n}_{11}) = 0$  and  $\Delta_2(\overleftarrow{n}_{11}) > 0$ . Note also that  $S_4$  and  $S_5$  hold false. Hence, it follows from Theorem 4 that using feedback in either transmitter-receiver pair does not improve the sum-rate, i.e.,  $\Sigma(\overleftarrow{n}_{11}) = 0$  and  $\Sigma(\overleftarrow{n}_{22}) = 0$ . These conclusions are observed in Figure 5, for the case  $\overleftarrow{n}_{11} = 4$  and  $\overleftarrow{n}_{22} = 0$ , where the capacity regions  $\mathcal{C}(0,0)$  (thick red line) and  $\mathcal{C}(4,0)$  (thin blue line) are plotted.

## V. IMPLICATIONS ON THE GAUSSIAN INTERFERENCE CHANNEL

Given a fixed tuple  $(\overline{SNR}_1, \overline{SNR}_2, INR_{12}, INR_{21})$ , let  $\underline{\mathcal{R}}(\overline{SNR}_1, \overline{SNR}_2)$  be the achievable region of the G-IC-NOF described by Theorem 2 in [17] with parameters  $\overline{SNR}_1$  and  $\overline{SNR}_2$ ; let  $\overline{\mathcal{R}}(\overline{SNR}_1, \overline{SNR}_2)$  be the converse region of the G-IC-NOF described by Theorem 3 in [17] with parameters  $\overline{SNR}_1$  and  $\overline{SNR}_2$ ; and let also  $\mathcal{C}(\overline{SNR}_1, \overline{SNR}_2)$  be the capacity region of the G-IC-NOF with parameters  $\overline{SNR}_1$  and  $\overline{SNR}_2$ . These regions satisfy the following inclusions:

$$\underline{\mathcal{R}}(\overleftarrow{\mathrm{SNR}}_1, \overleftarrow{\mathrm{SNR}}_2) \subseteq \mathcal{C}(\overleftarrow{\mathrm{SNR}}_1, \overleftarrow{\mathrm{SNR}}_2) \subseteq \overline{\mathcal{R}}(\overleftarrow{\mathrm{SNR}}_1, \overleftarrow{\mathrm{SNR}}_2).$$
(42)

#### A. Improvement Metrics

In order to quantify the benefits of channel-output feedback in enlarging the achievable region  $\underline{\mathcal{R}}(\overline{SNR}_1, \overline{SNR}_2)$  or the converse region  $\overline{\mathcal{R}}(\overline{SNR}_1, \overline{SNR}_2)$ , consider the following improvement metrics, which are similar to those defined in Sec. III-B for the LD-IC-NOF. The improvement metrics on the individual rates are defined as

$$\Delta_{1}^{A}(\overleftarrow{SNR}_{1},\overleftarrow{SNR}_{2}) = \max_{0 < R_{2} < R_{2}^{*}} \left\{ \sup \left\{ R_{1} : (R_{1}, R_{2}) \in \underline{\mathcal{R}}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) \right\} - \sup \left\{ R_{1}^{\dagger} : (R_{1}^{\dagger}, R_{2}) \in \underline{\mathcal{R}}(0, 0) \right\} \right\},$$
(43)



Fig. 5. Capacity regions C(0,0) (thick red line) and C(4,0) (thin blue line), with  $\vec{n}_{11} = 5$ ,  $\vec{n}_{22} = 1$ ,  $n_{12} = 3$ ,  $n_{21} = 4$ .

$$\Delta_{2}^{A}(\overleftarrow{SNR}_{1},\overleftarrow{SNR}_{2}) = \max_{0 < R_{1} < R_{1}^{*}} \left\{ \sup \left\{ R_{2} : (R_{1}, R_{2}) \in \underline{\mathcal{R}}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) \right\} - \sup \left\{ R_{2}^{\dagger} : (R_{1}, R_{2}^{\dagger}) \in \underline{\mathcal{R}}(0, 0) \right\} \right\},$$
(44)

$$\Delta_{1}^{C}(\overleftarrow{SNR}_{1},\overleftarrow{SNR}_{2}) = \max_{0 < R_{2} < R_{2}^{\dagger}} \left\{ \sup \left\{ R_{1} : (R_{1}, R_{2}) \in \overline{\mathcal{R}}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) \right\} - \sup \left\{ R_{1}^{\dagger} : (R_{1}^{\dagger}, R_{2}) \in \overline{\mathcal{R}}(0, 0) \right\} \right\}, \text{ and}$$
(45)

$$\Delta_{2}^{C}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) = \max_{0 < R_{1} < R_{1}^{\dagger}} \left\{ \sup \left\{ R_{2} : (R_{1}, R_{2}) \in \overline{\mathcal{R}}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) \right\} - \sup \left\{ R_{1}^{\dagger} : (R_{1}, R_{1}^{\dagger}) \in \overline{\mathcal{R}}(0, 0) \right\} \right\}$$
(46)

$$-\sup\left\{R_{2}^{\dagger}:(R_{1},R_{2}^{\dagger})\in\overline{\mathcal{R}}(0,0)\right\}\bigg\},$$
(46)

with

$$R_1^* = \sup \{ r_1 : (r_1, r_2) \in \underline{\mathcal{R}}(0, 0) \},$$
(47)

$$R_{2}^{*} = \sup \left\{ r_{2} : (r_{1}, r_{2}) \in \underline{\mathcal{R}}(0, 0) \right\},$$
(48)

$$R_1^{\dagger} = \sup \{ r_1 : (r_1, r_2) \in \overline{\mathcal{R}}(0, 0) \}, \text{ and}$$
 (49)

$$R_{2}^{\dagger} = \sup \left\{ r_{2} : (r_{1}, r_{2}) \in \overline{\mathcal{R}}(0, 0) \right\}.$$
(50)

Alternatively, the maximum improvements of the sum-rate  $\Sigma^{A}(\overline{SNR}_{1}, \overline{SNR}_{2})$  and  $\Sigma^{C}(\overline{SNR}_{1}, \overline{SNR}_{2})$  with respect to the case without feedback are:

$$\Sigma^{A}(\overleftarrow{SNR}_{1},\overleftarrow{SNR}_{2}) = \sup \left\{ R_{1} + R_{2} : (R_{1}, R_{2}) \in \underline{\mathcal{R}}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) \right\} - \sup \left\{ R_{1}^{\dagger} + R_{2}^{\dagger} : (R_{1}^{\dagger}, R_{2}^{\dagger}) \in \underline{\mathcal{R}}(0, 0) \right\}, \text{ and } (51)$$
$$\Sigma^{C}(\overleftarrow{SNR}_{1}, \overleftarrow{SNR}_{2}) =$$

$$\sup \left\{ R_1 + R_2 : (R_1, R_2) \in \overline{\mathcal{R}}(\overleftarrow{\mathrm{SNR}}_1, \overleftarrow{\mathrm{SNR}}_2) \right\} - \sup \left\{ R_1^{\dagger} + R_2^{\dagger} : (R_1^{\dagger}, R_2^{\dagger}) \in \overline{\mathcal{R}}(0, 0) \right\}.$$
(52)

## B. Approximate Thresholds on the Feedback SNRs

In Section II-C, the connections between the LD-IC-NOF and the G-IC-NOF discussed. Using were these connections, a G-IC with fixed parameters  $\left(\overline{\text{SNR}}_1, \overline{\text{SNR}}_2, \text{INR}_{12}, \text{INR}_{21}\right)$ approximated is by with parameters  $\vec{n}_{11} = |\frac{1}{2}\log_2(\overline{\text{SNR}}_1)|$ , an LD-IC  $\overrightarrow{n}_{22} = \lfloor \frac{1}{2} \log_2(\overrightarrow{\text{SNR}}_2) \rfloor, \quad n_{12} = \lfloor \frac{1}{2} \log_2(\text{INR}_{12}) \rfloor, \text{ and }$  $n_{21} = \lfloor \frac{1}{2} \log_2(\text{INR}_{21}) \rfloor$ . From this observation, the results from Theorem 1 - Theorem 4 can be used to determine the feedback SNR thresholds beyond which either an individual rate or the sum-rate is improved in the original G-IC-NOF. The procedure consists on using the equalities  $\overleftarrow{n}_{ii} = \lfloor \frac{1}{2} \log_2 (SNR_i) \rfloor$ , with  $i \in \{1, 2\}$ . Hence, the corresponding thresholds in the G-IC can be approximated by:

$$\overleftarrow{\mathrm{SNR}}_{i}^{*} = 2^{2\overleftarrow{n}_{ii}^{*}}, \qquad (53a)$$

$$\overline{\mathrm{SNR}}_{i}^{\dagger} = 2^{2\overline{n}_{ii}^{\dagger}}, \text{ and}$$
 (53b)

$$\overleftarrow{\mathrm{SNR}}_{i}^{+} = 2^{2\overleftarrow{n}_{ii}^{+}}.$$
(53c)

When the corresponding LD-IC-NOF is such that its capacity region can be improved when  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^*$  (Theorem 1), for a given  $i \in \{1, 2\}$ , it is expected that either the achievability or converse regions of the original G-IC-NOF become larger when  $\overline{SNR}_i > \overline{SNR}_i^*$ . Similarly, when the corresponding LD-IC-NOF is such that  $\Delta_i(\overleftarrow{n}_{ii}) > 0$  or  $\Delta_i(\overleftarrow{n}_{ij}) > 0$ , it is expected to observe an improvement on the individual rate  $R_i$  by either using feedback in transmitter-receiver pair i, with  $\overline{SNR}_i > \overline{SNR}_i^{\dagger}$  or by using feedback in transmitterreceiver pair j, with  $SNR_i > SNR_i^*$ . In the case of the sum-rate, when the corresponding LD-IC-NOF is such that  $\Sigma(\overleftarrow{n}_{ii}) > 0$  using feedback in transmitter-receiver pair *i*, with  $\overleftarrow{n}_{ii} > \overleftarrow{n}_{ii}^+$ , (Theorem 4), it is expected to observe an improvement on the sum-rate by using feedback in transmitterreceiver pair i, with  $\hat{S}NR_i > \hat{S}NR_i^+$ . Finally, when no improvement in a given metric is observed in the LD-IC-NOF, i.e.,  $\Delta_1(\overleftarrow{n}_{11}) = 0$ ,  $\Delta_1(\overleftarrow{n}_{22}) = 0$ ,  $\Delta_2(\overleftarrow{n}_{11}) = 0$ ,  $\Delta_2(\overleftarrow{n}_{22}) = 0, \ \Sigma(\overleftarrow{n}_{11}) = 0, \ \text{or} \ \Sigma(\overleftarrow{n}_{22}) = 0, \ \text{only a negli-}$ gible improvement (if any) is observed in the corresponding metric of the G-IC-NOF. For instance, when  $\Delta_1(\overleftarrow{n}_{11}) = 0$ , it is expected that  $\Delta_1^{C}(\overleftarrow{SNR}_1, 0) < \epsilon$  and  $\Delta_1^{A}(\overleftarrow{SNR}_1, 0) < \epsilon$ , with  $\epsilon > 0$ . Similarly, when  $\Delta_2(\overline{n}_{11}) = 0$ , it is expected that  $\Delta_2^{\rm C}(\mathrm{\dot{S}NR}_1, 0) < \epsilon$  and  $\Delta_2^{\rm A}(\mathrm{\dot{S}NR}_1, 0) < \epsilon$ . Finally, when  $\Sigma(\overleftarrow{n}_{11}) = 0$ , it is expected that  $\Sigma^{C}(\overleftarrow{SNR}_{1}, 0) < \epsilon$  and  $\Sigma^{A}(\overline{SNR}_{1}, 0) < \epsilon.$ 

#### C. Examples

The following examples highlight the relevance of the approximations in (53).

Example 4: Consider a G-IC with parameters  $\overrightarrow{SNR}_1 = 44 \text{dB}$ ,  $\overrightarrow{SNR}_2 = 44 \text{dB}$ ,  $\overrightarrow{INR}_{12} = 20 \text{dB}$ , and  $\overrightarrow{INR}_{21} = 33 \text{dB}$ .

The linear deterministic approximation to the G-IC in Example 4 is the one presented in Example 1. Hence,  $\overline{n}_{11}^* = \overline{n}_{11}^\dagger = \overline{n}_{11}^+ = 5$  and  $\overline{n}_{22}^* = \overline{n}_{22}^\dagger = \overline{n}_{22}^+ = 3$ .



Fig. 6. Improvement metrics  $\Delta_i^A$ ,  $\Delta_i^C$ ,  $\Sigma^A$ , and  $\Sigma^C$  as functions of  $\overline{SNR}_1$  and  $\overline{SNR}_2$ , with  $i \in \{1, 2\}$ , for Example 4.

This implies that  $\overleftarrow{SNR}_1^* = \overleftarrow{SNR}_1^\dagger = \overrightarrow{SNR}_1^+ = 30 dB$  and  $\overrightarrow{SNR}_2^* = \overrightarrow{SNR}_2^\dagger = \overrightarrow{SNR}_2^\dagger = 18 dB.$ 

Figure 6 shows that significant improvements on the metrics  $\Delta_i^{\text{A}}(\dot{\text{SNR}}_1, \dot{\text{SNR}}_2), \ \Delta_i^{\text{C}}(\dot{\text{SNR}}_1, \dot{\text{SNR}}_2), \ \Sigma^{\text{A}}(\dot{\text{SNR}}_1, \dot{\text{SNR}}_2)$  and  $\Sigma^{\text{C}}(\dot{\text{SNR}}_1, \dot{\text{SNR}}_2)$  are obtained when the feedback SNRs

are beyond the corresponding thresholds. More importantly, negligible effects are observed when  $\overline{SNR}_1 < \overline{SNR}_1^*$  and  $\overline{SNR}_2 < \overline{SNR}_2^*$ .

 $\begin{array}{cccc} \underline{Example} & 5: & \underline{Consider} & a & \underline{G\text{-}IC} & with & parameters \\ \overline{\mathrm{SNR}}_1 = 33\mathrm{dB}, & \overline{\mathrm{SNR}}_2 & = & 9\mathrm{dB}, & \mathrm{INR}_{12} & = & 20\mathrm{dB}, & and \end{array}$ 

 $INR_{21} = 27 dB.$ 

The linear deterministic approximation to the G-IC in Example 5 is the one presented in Example 3. Hence,  $\overline{m}_{11}^* = 3$ , which implies that  $SNR_1^* = 18$ dB. It follows from the LD-IC that using feedback in transmitter-receiver pair 1 exclusively increases the individual rate  $R_2$ . This is observed in Figure 7c. Note that the improvement in the individual rate  $R_2$  for all  $\overline{SNR}_1 < \overline{SNR}_1^*$  is negligible. Significant improvement is observed only beyond the threshold  $\overline{SNR}_1^*$ .

Note also that using feedback in either transmitterreceiver pair does not improve the rate  $R_1$  in the LD-IC-NOF, i.e.,  $\Delta_1(\overleftarrow{n}_{11}) = \Delta_1(\overleftarrow{n}_{22}) = 0$ . This is also verified in the G-IC-NOF by Figure 7a, Figure 7b, and Figure 7d, where  $\Delta_1^A \left(-100 \text{dB}, \overline{\text{SNR}}_2\right) < 0.15$  and  $\Delta_1^{\rm C} \left(-100 \mathrm{dB}, \overline{\mathrm{SNR}}_2\right) < 0.1.$ 

Finally, note that using feedback in either transmitterreceiver pair does not increase the sum-rate in the LD-IC-NOF, i.e.,  $\Sigma(\overline{n}_{11}) = \Sigma(\overline{n}_{22}) = 0$ . This is also verified in the G-IC-NOF by Figure 7e and Figure 7f, where  $\Sigma^{A}(\overline{SNR}_{1}, -100 \text{dB})$ <0.15,  $\Sigma^{C}(\overline{SNR}_{1}, -100 dB) < 0.05, \Sigma^{A}(-100 dB, \overline{SNR}_{2}) < 0.15,$ and  $\Sigma^{C}$   $\left(-100 \text{dB}, \overline{\text{SNR}}_{2}\right) < 0.05$ .

#### VI. ACKNOWLEDGMENTS

The authors gratefully acknowledge the constructive comments and suggestions made by the editor and the reviewers aiming to improve the clarity of this paper.

#### VII. CONCLUSIONS

In this paper, for any 4-tuple  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}) \in \mathbb{N}^4$ , the exact values of the feedback parameters  $\overleftarrow{n}_{11}$  and  $\overleftarrow{n}_{22}$ of the two-user LD-IC-NOF beyond which the capacity region enlarges are characterized. That is, the exact values of  $\overline{n}_{11}$  (resp.  $\overline{n}_{22}$ ) for which  $\mathcal{C}(0,0) \subset \mathcal{C}(\overline{n}_{11},0)$  (resp.  $\mathcal{C}(0,0) \subset \mathcal{C}(0,\overline{n}_{22})$  holds with strict inclusion. The SNRs in the feedback links beyond which feedback plays a significant role in terms of increasing the individual rates or the sumrate in the G-IC are also identified. The relevance of this work lies on the fact that it allows identifying a number of scenarios in any G-IC for which one of the following statements is true: (a) feedback does not enlarge the capacity region; (b) feedback enlarges the capacity region and the sumrate is greater than the largest sum-rate without feedback; and (c) feedback enlarges the capacity region but no significant improvement is observed in the sum-rate.

#### APPENDIX A

# **PROOF OF THEOREM 1: ENLARGEMENT OF THE CAPACITY REGION BY USING FEEDBACK IN ONE TRANSMITTER-RECEIVER PAIR**

The proof of Theorem 1 is obtained by comparing  $\mathcal{C}(\overline{n}_{11},0)$  (resp.  $\mathcal{C}(0,\overline{n}_{22})$ ) and  $\mathcal{C}(0,0)$ , with fixed parameters  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ , and  $n_{21}$ . More specifically, for each tuple  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21})$ , the exact value  $\overleftarrow{n}_{11}^*$  (resp  $\overleftarrow{n}_{22}^*$ ) for which any  $\overleftarrow{n}_{11} > \overleftarrow{n}_{11}^*$  (resp  $\overleftarrow{n}_{22} > \overleftarrow{n}_{22}^*$ ) ensures  $\mathcal{C}(0,0) \subset \mathcal{C}(\overleftarrow{n}_{11},0)$  (resp.  $\mathcal{C}(0,0) \subset \mathcal{C}(0,\overleftarrow{n}_{22})$ ) is calculated. This procedure is tedious and repetitive, and thus, in this appendix only one combination of interference regimes is studied, namely, VWIR - VWIR.

#### Proof:

Consider that both transmitter-receiver pairs are in VWIR, that is.

$$\alpha_1 = \frac{n_{12}}{\overrightarrow{n}_{11}} \leqslant \frac{1}{2} \text{ and } \alpha_2 = \frac{n_{21}}{\overrightarrow{n}_{22}} \leqslant \frac{1}{2}.$$
(54)

When the conditions in (54) are fulfilled, it follows from Theorem 1 in [17] that  $\mathcal{C}(0,0)$  is the set of non-negative rate pairs  $(R_1, R_2) \in \mathbb{R}^2_+$  that satisfy:

$$R_1 \leqslant \overrightarrow{n}_{11} \triangleq \theta_1, \tag{55a}$$

$$R_2 \leqslant \overrightarrow{n}_{22} \triangleq \theta_2, \tag{55b}$$

 $R_1 + R_2 \leq \min\left(\max\left(\overrightarrow{n}_{22}, n_{12}\right) + \overrightarrow{n}_{11} - n_{12}, \right)$ 

$$\max\left(\vec{n}_{11}, n_{21}\right) + \vec{n}_{22} - n_{21} \triangleq \theta_3, \qquad (55c)$$

$$R_1 + R_2 \leqslant \max\left(\vec{n}_{11} - n_{12}, n_{21}\right) + \max\left(\vec{n}_{22} - n_{21}, n_{12}\right) \\ \triangleq \theta_4, \tag{55d}$$

$$2R_1 + R_2 \leq \max\left(\vec{n}_{11}, n_{21}\right) + \vec{n}_{11} - n_{12} + \max\left(\vec{n}_{22} - n_{21}, n_{12}\right) \triangleq \theta_5,$$
(55e)

$$R_{1} + 2R_{2} \leqslant \max\left(\vec{n}_{22}, n_{12}\right) + \vec{n}_{22} - n_{21} + \max\left(n_{21}, \vec{n}_{11} - n_{12}\right) \triangleq \theta_{6}.$$
(55f)

 $\mathbb{N}^5$ Note that for all  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{22}, n_{12}, n_{21}, \overleftarrow{n}_{22})$  $\in$  $\max(\overrightarrow{n}_{11}, n_{12}), \text{ it follows}$ and  $\overleftarrow{n}_{11}$ >that  $\mathcal{C}(\overleftarrow{n}_{11}, \overleftarrow{n}_{22}) = \mathcal{C}(\max(\overrightarrow{n}_{11}, n_{12}), \overleftarrow{n}_{22}).$  Hence, in the following, the analysis is restricted to the following condition:

$$\overleftarrow{n}_{11} \leqslant \max\left(\overrightarrow{n}_{11}, n_{12}\right). \tag{56}$$

Under conditions (54) and (56), it follows from Theorem 1 in [17] that  $\mathcal{C}(\overleftarrow{n}_{11}, 0)$  is the set of rate pairs  $(R_1, R_2) \in \mathbb{R}^2_+$ that satisfy:

$$R_1 \leqslant \overrightarrow{n}_{11}, \tag{57a}$$

$$R_2 \leqslant \overrightarrow{n}_{22},\tag{57b}$$

$$R_1 + R_2 \leq \min\left(\max\left(\overrightarrow{n}_{22}, n_{12}\right) + \overrightarrow{n}_{11} - n_{12}, \right)$$

$$\max(\vec{n}_{11}, n_{21}) + \vec{n}_{22} - n_{21}), \qquad (57c)$$

$$R_{1} + R_{2} \leqslant \max\left(\vec{n}_{11} - n_{12}, n_{21}, \overleftarrow{n}_{11}\right) + \max\left(\vec{n}_{22} - n_{21}, n_{12}\right) \triangleq \theta_{7},$$
(57d)

$$2R_1 + R_2 \leqslant \max\left(\vec{n}_{11}, n_{21}\right) + \vec{n}_{11} - n_{12} + \max\left(\vec{n}_{22} - n_{21}, n_{12}\right),$$
(57e)

$$R_{1} + 2R_{2} \leqslant \max\left(\vec{n}_{22}, n_{12}\right) + \vec{n}_{22} - n_{21} + \max\left(\vec{n}_{11} - n_{12}, n_{21}, \overleftarrow{n}_{11}\right) \triangleq \theta_{8}, \quad (57f)$$

pomparing 
$$C(0,0)$$
 and  $C(\overleftarrow{n}_{11},0)$ , note that (55a), (55b),  
nd (55e) are equivalent to (57a), (57b), (57c), and

When co (55c), a (57e), respectively. That being the case, the region  $\mathcal{C}(\overleftarrow{n}_{11}, 0)$ is greater than the region  $\mathcal{C}(0,0)$  if at least one of the following conditions holds true:

$$\min(\theta_3, \theta_4, \theta_1 + \theta_2, \theta_5, \theta_6) < \theta_7 < \min(\theta_3, \theta_1 + \theta_2, \theta_5, \theta_8),$$
(58a)



Fig. 7. Improvement metrics  $\Delta_i^A$ ,  $\Delta_i^C$ ,  $\Sigma^A$ , and  $\Sigma^C$  as functions of  $\overline{SNR}_1$  and  $\overline{SNR}_2$ , with  $i \in \{1, 2\}$ , for Example 5.

$$\min(\theta_6, \theta_1 + 2\theta_2, \theta_2 + \theta_3, \theta_4 + \theta_2) < \theta_8 < \min(\theta_1 + 2\theta_2, \theta_2 + \theta_3, \theta_2 + \theta_7).$$
(58b)

Condition (58a) implies that the active sum-rate bound in  $\mathcal{C}(\overline{n}_{11}, 0)$  is greater than the active sum-rate bound in  $\mathcal{C}(0, 0)$ . Condition (58b) implies that the active weighted sum-rate bound on  $R_1 + 2R_2$  in  $\mathcal{C}(\overleftarrow{n}_{11}, 0)$  is greater than the active weighted sum-rate bound on  $R_1 + 2R_2$  in C(0, 0).

To simplify the inequalities containing the operator  $\max(\cdot, \cdot)$  in (57) and (55), the following 4 cases are identified:

Case  $1: \overrightarrow{n}_{11} - n_{12} < n_{21}$  and  $\overrightarrow{n}_{22} - n_{21} < n_{12}$ ; (59)

- Case  $2: \overrightarrow{n}_{11} n_{12} < n_{21}$  and  $\overrightarrow{n}_{22} n_{21} \ge n_{12}$ ; (60)
- Case 3:  $\vec{n}_{11} n_{12} \ge n_{21}$  and  $\vec{n}_{22} n_{21} < n_{12}$ ; and (61) 2)

Case 4: 
$$\vec{n}_{11} - n_{12} \ge n_{21}$$
 and  $\vec{n}_{22} - n_{21} \ge n_{12}$ . (62)

Case 1: Under condition (54), the Case 1, i.e., (59), is not

possible.

<u>Case 2</u>: Under condition (54), the case 2, i.e., (60), this case is possible.

Plugging (60) into (57) yields:

$$R_{1} + R_{2} \leqslant \min\left(\vec{n}_{22} + \vec{n}_{11} - n_{12}, \max\left(\vec{n}_{11}, n_{21}\right) + \vec{n}_{22} - n_{21}\right),$$
(63a)

$$R_1 + R_2 \leq \max(n_{21}, \overleftarrow{n}_{11}) + \overrightarrow{n}_{22} - n_{21},$$
 (63b)

$$R_1 + 2R_2 \leqslant 2\vec{n}_{22} - n_{21} + \max(n_{21}, \overleftarrow{n}_{11}).$$
 (63c)

Plugging (60) into (55) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{22}, \tag{64a}$$

$$R_1 + 2R_2 \leqslant 2 \vec{n}_{22}.$$
 (64b)

To simplify the inequalities containing the operator  $\max(\cdot, \cdot)$  in (63), the following 2 cases are identified:

Case 
$$2a: \vec{n}_{11} > n_{21};$$
 and (65)

Case 
$$2\mathbf{b}: \overrightarrow{n}_{11} \leqslant n_{21}$$
. (66)

Case 2a: Plugging (65) into (63) yields:

$$R_1 + R_2 \leqslant \vec{n}_{11} + \vec{n}_{22} - n_{21}, \tag{67a}$$

$$R_1 + R_2 \leqslant \max(n_{21}, n_{11}) + n_{22} - n_{21}, \quad (6/b)$$
  
$$R_1 + 2R_2 \leqslant 2 \overrightarrow{n}_{22} - n_{21} + \max(n_{21}, \overleftarrow{n}_{11}). \quad (67c)$$

Comparing inequalities (67a) and (67b) with inequality (64a),  
it can be verified that 
$$\min\left(\vec{n}_{11} + \vec{n}_{22} - n_{21}, \max\left(n_{21}, \vec{n}_{11}\right) + \vec{n}_{22} - n_{21}\right) > \vec{n}_{22}$$
, i.e., condition (58a) holds,  
when  $\overleftarrow{n}_{11} > n_{21}$ . Comparing inequalities (67c) and (64b),  
it can be verified that  $2\vec{n}_{22} - n_{21} + \max\left(n_{21}, \overleftarrow{n}_{11}\right) > 2\vec{n}_{22}$ ,  
i.e., condition (58b) holds, when  $\overleftarrow{n}_{11} > n_{21}$ . Therefore,  
 $\overleftarrow{n}_{11}^* = n_{21}$  under conditions (54), (56), (60), and (65).

Case 2b: Plugging (66) into (63) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{22},\tag{68a}$$

$$R_1 + R_2 \leq \max(n_{21}, \overleftarrow{n}_{11}) + \overrightarrow{n}_{22} - n_{21},$$
 (68b)

$$R_1 + 2R_2 \leqslant 2 \overrightarrow{n}_{22} - n_{21} + \max(n_{21}, \overleftarrow{n}_{11}).$$
 (68c)

Comparing inequalities (68a) and (68b) with inequality (64a), it can be verified that min  $(\overrightarrow{n}_{22}, \max(n_{21}, \overleftarrow{n}_{11}) + \overrightarrow{n}_{22} - n_{21}) = \overrightarrow{n}_{22}$ , i.e., condition (58a) does not hold, for all  $\overleftarrow{n}_{11} \in \mathbb{N}$ . Comparing inequalities (68c) and (64b) it can be verified that  $2\overrightarrow{n}_{22}-n_{21}+\max(n_{21}, \overleftarrow{n}_{11}) > 2\overrightarrow{n}_{22}$ , when  $\overleftarrow{n}_{11} > n_{21}$ , which implies that  $\overleftarrow{n}_{11} > \max(\overrightarrow{n}_{11}, n_{12})$ . However, under the conditions (54), (56), (60), and (66), the bounds (64b) and (68c) are not active. Hence, condition (58b) does not hold. Therefore, for all  $\overleftarrow{n}_{11} \in \mathbb{N}$ , the capacity region cannot be enlarged under conditions (54), (56), (60), and (66).

<u>Case 3:</u> Under condition (54), the Case 3, i.e., (61), is possible. Plugging (61) into (57) yields:

$$R_{1} + R_{2} \leqslant \min\left(\max\left(\vec{n}_{22}, n_{12}\right) + \vec{n}_{11} - n_{12}, \vec{n}_{11} + \vec{n}_{22} - n_{21}\right),$$
(69a)

$$R_1 + R_2 \leq \max\left(\overrightarrow{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right) + n_{12}, \tag{69b}$$

$$R_{1} + 2R_{2} \leqslant \max(\overrightarrow{n}_{22}, n_{12}) + \overrightarrow{n}_{22} - n_{21} + \max(\overrightarrow{n}_{11} - n_{12}, \overleftarrow{n}_{11}).$$
(69c)

Plugging (61) into (55) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{11}, \tag{70a}$$

$$R_1 + 2R \leqslant \max(\overrightarrow{n}_1 - n_1) + \overrightarrow{n}_2 - n_2 + \overrightarrow{n}_2 - n_3 \tag{70b}$$

$$R_1 + 2R_2 \leqslant \max(\vec{n}_{22}, n_{12}) + \vec{n}_{22} - n_{21} + \vec{n}_{11} - n_{12}.$$
(70b)

To simplify the inequalities containing the operator  $\max(\cdot, \cdot)$  in (69) and (70), the following 2 cases are identified:

Case 
$$3a: \vec{n}_{22} > n_{12};$$
 and (71)

Case 3b: 
$$\overrightarrow{n}_{22} \leq n_{12}$$
. (72)

Case 3a: Plugging (71) into (69) yields:

$$R_1 + R_2 \leqslant \vec{n}_{22} + \vec{n}_{11} - n_{12}, \tag{73a}$$

$$R_1 + R_2 \leq \max\left(\vec{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right) + n_{12},\tag{73b}$$

$$R_1 + 2R_2 \leqslant 2\,\overline{n}_{22} - n_{21} + \max\left(\overline{n}_{11} - n_{12},\,\overline{n}_{11}\right). \quad (73c)$$

Plugging (71) into (70) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{11},\tag{74a}$$

$$R_1 + 2R_2 \leqslant 2\vec{n}_{22} - n_{21} + \vec{n}_{11} - n_{12}.$$
 (74b)

Comparing inequalities (73a) and (73b) with inequality (74a), it can be verified that min  $(\vec{n}_{22} + \vec{n}_{11} - n_{12}, \max(\vec{n}_{11} - n_{12}, (\vec{n}_{11}) + n_{12}) > \vec{n}_{11}$ , i.e., condition (58a) holds, when  $(\vec{n}_{11} > \vec{n}_{11} - n_{12})$ . Comparing inequalities (73c) and (74b), it can be verified that  $2\vec{n}_{22} - n_{21} + \max(\vec{n}_{11} - n_{12}, (\vec{n}_{11}) > 2\vec{n}_{22} - n_{21} + \vec{n}_{11} - n_{12})$ , i.e., condition (58b) holds, when  $(\vec{n}_{11} > \vec{n}_{11} - n_{12})$ . Therefore,  $(\vec{n}_{11}^* = \vec{n}_{11} - n_{12})$  under conditions (54), (56), (61), and (71).

Case 3b: Plugging (72) into (69) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{11}, \tag{75a}$$

$$R_1 + R_2 \leqslant \max\left(\overrightarrow{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right) + n_{12},$$
 (75b)

$$R_1 + 2R_2 \leq n_{12} + \vec{n}_{22} - n_{21} + \max\left(\vec{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right). (75c)$$

Plugging (71) into (70) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{11}, \tag{76a}$$

$$R_1 + 2R_2 \leqslant \overrightarrow{n}_{22} - n_{21} + \overrightarrow{n}_{11}. \tag{76b}$$

Comparing inequalities (75a) and (75b) with inequality (76a), it can be verified that  $\min\left(\vec{n}_{11}, \max\left(\vec{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right) + n_{12}\right) = \vec{n}_{11}$ , i.e., condition (58a) does not hold, for all  $\overleftarrow{n}_{11} \in \mathbb{N}$ . Comparing inequalities (75c) and (76b), it can be verified that  $n_{12} + \vec{n}_{22} - n_{21} + \max\left(\vec{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right) > \vec{n}_{22} - n_{21} + \vec{n}_{11}$ , i.e., condition (58b) holds, when  $\overleftarrow{n}_{11} > \vec{n}_{11} - n_{12}$ . Therefore,  $\overleftarrow{n}_{11}^* = \vec{n}_{11} - n_{12}$  under conditions (54), (56), (61), and (72).

<u>Case 4:</u> Under condition (54), Case 4, i.e., (62), is possible. Plugging (62) into (57) yields:

$$R_{1} + R_{2} \leqslant \min\left(\vec{n}_{22} + \vec{n}_{11} - n_{12}, \vec{n}_{11} + \vec{n}_{22} - n_{21}\right), (77a)$$

$$R_{1} + R_{2} \leqslant \max\left(\vec{n}_{11} - n_{12}, \overleftarrow{n}_{11}\right) + \vec{n}_{22} - n_{21}, \quad (77b)$$

$$R_1 + 2R_2 \leqslant 2 \,\overline{n}_{22} - n_{21} + \max\left(\overline{n}_{11} - n_{12}, \,\overline{n}_{11}\right). \tag{77c}$$

Plugging (62) into (55) yields:

$$R_1 + R_2 \leqslant \overrightarrow{n}_{11} - n_{12} + \overrightarrow{n}_{22} - n_{21}, \tag{78a}$$

$$R_1 + 2R_2 \leqslant 2\vec{n}_{22} - n_{21} + \vec{n}_{11} - n_{12}.$$
 (78b)

Comparing inequalities (77a) and (77b) with inequality (78a), it can be verified that  $\min(\min(\overrightarrow{n}_{22} + \overrightarrow{n}_{11} - n_{12}, \overrightarrow{n}_{11} +$ 

 $\vec{n}_{22} - n_{21}$ , max  $(\vec{n}_{11} - n_{12}, \vec{n}_{11}) + \vec{n}_{22} - n_{21}$  >  $\vec{n}_{11} - n_{12}$  $n_{12} + \overrightarrow{n}_{22} - n_{21}$ , i.e., condition (58a) holds, when  $\overleftarrow{n}_{11} >$  $\vec{n}_{11} - n_{12}$ . Comparing inequalities (77c) and (78b), it can be verified that:  $2\vec{n}_{22} - n_{21} + \max(\vec{n}_{11} - n_{12}, \overleftarrow{n}_{11}) >$  $2\vec{n}_{22} - n_{21} + \vec{n}_{11} - n_{12}$ , i.e., condition (58b) holds, when  $\overleftarrow{n}_{11} > \overrightarrow{n}_{11} - n_{12}.$ Therefore,  $\overleftarrow{n}_{11}^* = \overrightarrow{n}_{11} - n_{12}$  under conditions (54), (56),

and (62).

From all the observations above, when both transmitterreceiver pairs are in VWIR (event  $E_1$  in (17) holds true), it follows that when  $\overleftarrow{n}_{11} > \overleftarrow{n}_{11}^*$  and  $\overrightarrow{n}_{11} > n_{21}$ (event  $E_{8,1}$  in (24) with i = 1 holds true) with  $\overleftarrow{n}_{11}^* =$  $\max(\overrightarrow{n}_{11} - n_{12}, n_{21})$ , then  $\mathcal{C}(0, 0) \subset \mathcal{C}(\overleftarrow{n}_{11}, 0)$ . Otherwise,  $\mathcal{C}(0,0) = \mathcal{C}(\overleftarrow{n}_{11},0)$ . Note that when events  $E_1$  and  $E_{8,1}$  hold simultaneously true, then the event  $S_{1,1}$  in (28) with i = 1holds true, which verifies the statement of Theorem 1. The same procedure can be applied for all the other combinations of interference regimes. This completes the proof.

#### APPENDIX B

## **PROOF OF THEOREM 2: IMPROVEMENT OF THE** INDIVIDUAL RATE $R_i$ by USING FEEDBACK IN LINK i

The proof of Theorem 2 is obtained by comparing  $\mathcal{C}(\overleftarrow{n}_{11},0)$  (resp.  $\mathcal{C}(0,\overleftarrow{n}_{22})$ ) and  $\mathcal{C}(0,0)$ , for all possible parameters  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ ,  $n_{21}$ , and  $\overleftarrow{n}_{11}$  (resp.  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ ,  $n_{21}$ , and  $\overleftarrow{n}_{22}$ ). More specifically, for each tuple  $(\overrightarrow{n}_{11}, \overrightarrow{n}_{12}, \overrightarrow{n}_{21}, \overrightarrow{n}_{21}, \overrightarrow{n}_{22})$  $\overrightarrow{n}_{22}, n_{12}, n_{21}$ , the exact value  $\overleftarrow{n}_{11}^{\dagger}$  (resp  $\overleftarrow{n}_{22}^{\dagger}$ ) for which any  $\overleftarrow{n}_{11} > \overleftarrow{n}_{11}^{\dagger}$  (resp  $\overleftarrow{n}_{22} > \overleftarrow{n}_{22}^{\dagger}$ ) ensures an improvement on  $R_1$  (resp.  $R_2$ ), i.e.,  $\Delta_1(\overleftarrow{n}_{11}, 0) > 0$  (resp.  $\Delta_2(0, \overleftarrow{n}_{22}) > 0$ ), is calculated. This procedure is tedious and repetitive, and thus, in this appendix only one combination of interference regimes is studied, namely, VWIR - VWIR.

Proof:

Consider that both transmitter-receiver pairs are in VWIR, i.e., conditions (54) hold. Under these conditions, the capacity regions  $\mathcal{C}(0,0)$  and  $\mathcal{C}(\overline{n}_{11},0)$  are given by (55) and (57), respectively. When comparing  $\mathcal{C}(0,0)$  and  $\mathcal{C}(\overleftarrow{n}_{11},0)$ , note that (55a), (55b), (55c), and (55e) are equivalent to (57a), (57b), (57c), and (57e), respectively. In this case any improvement on  $R_1$  is produced by an improvement on  $R_1 + R_2$  (condition (58a)) or  $2R_1 + R_2$  (condition (58a)), and thus, the proof of Theorem 2 in these particular interference regimes follows exactly the same steps as in Theorem 1. This completes the proof.

#### APPENDIX C

# **PROOF OF THEOREM 4: IMPROVEMENT OF THE SUM-RATE** CAPACITY BY USING FEEDBACK IN ONE **TRANSMITTER-RECEIVER PAIR**

The proof of Theorem 4 is obtained by comparing  $\mathcal{C}(\overline{n}_{11},0)$  (resp.  $\mathcal{C}(0,\overline{n}_{22})$ ) and  $\mathcal{C}(0,0)$ , for all possible parameters  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ ,  $n_{21}$ , and  $\overleftarrow{n}_{11}$  (resp.  $\vec{n}_{11}$ ,  $\vec{n}_{22}$ ,  $n_{12}$ ,  $n_{21}$ , and  $\overleftarrow{n}_{22}$ ). More specifically, for each tuple ( $\overrightarrow{n}_{11}$ ,  $\overrightarrow{n}_{22}, n_{12}, n_{21}$ , the exact value  $\overleftarrow{n}_{11}^+$  (resp  $\overleftarrow{n}_{22}^+$ ) for which any  $\overleftarrow{n}_{11} > \overleftarrow{n}_{11}^+$  (resp  $\overleftarrow{n}_{22} > \overleftarrow{n}_{22}^+$ ) ensures an improvement on  $R_1 + R_2$ , i.e.,  $\Sigma(\overline{n}_{11}, 0) > 0$  (resp.  $\Sigma(0, \overline{n}_{22}) > 0$ ), is calculated. This procedure is tedious and repetitive, and thus, in this appendix only one combination of interference regimes is studied, namely, VWIR - VWIR.

#### Proof:

Consider that both transmitter-receiver pairs are in VWIR, i.e., conditions (54) hold. Under these conditions, the capacity regions  $\mathcal{C}(0,0)$  and  $\mathcal{C}(\overleftarrow{n}_{11},0)$  are given by (55) and (57), respectively. When comparing  $\mathcal{C}(0,0)$  and  $\mathcal{C}(\overleftarrow{n}_{11},0)$ , note that (55a), (55b), (55c), and (55e) are equivalent to (57a), (57b), (57c), and (57e), respectively.

In this case, the proof is focused on any improvement on  $R_1 + R_2$  (condition (58a)), and thus, the proof of Theorem 4 in these particular interference regimes follows exactly the same steps as in Theorem 1.

From the analysis presented in Appendix A, it follows that: <u>Case 2a</u>: condition (58a) holds true, when  $\overleftarrow{n}_{11} > n_{21}$  under conditions (54), (56), (60), and (65).

Case 2b: condition (58a) does not hold true, under conditions (54), (60), and (66).

Case 3*a*: condition (58a) holds true, when  $\overleftarrow{n}_{11} > \overrightarrow{n}_{11} - n_{12}$ under conditions (54), (56), (61), and (71).

Case 3b: condition (58a) does not hold true, when  $\overleftarrow{n}_{11} > \overrightarrow{n}_{11} - n_{12}$  under conditions (54), (56), (61), and (72). Case 4: condition (58a) holds true, when  $\overleftarrow{n}_{11} > \overrightarrow{n}_{11} - n_{12}$ under conditions (54), (56), and (62).

From all the observations above, when both transmitterreceiver pairs are in VWIR (event  $E_1$  in (17) holds true), it follows that when  $\overleftarrow{n}_{11} > \overleftarrow{n}_{11}^+$ ,  $\overrightarrow{n}_{11} > n_{21}$  (event  $E_{8,1}$  in (24) with i = 1 holds true),  $\overrightarrow{n}_{22} > n_{12}$  (event  $E_{8,2}$  in (24) with i = 2 holds true),  $\overrightarrow{n}_{11} + \overrightarrow{n}_{22} > n_{12} + 2n_{21}$  (event  $E_{10,1}$  in (26) with i = 1 holds true), and  $\overrightarrow{n}_{11} + \overrightarrow{n}_{22} > 1$  $n_{21} + 2n_{12}$  (event  $E_{10,2}$  in (26) with i = 2 holds true) with  $\overline{n}_{11}^+ = \max(\overline{n}_{11} - n_{12}, n_{21})$ , then  $\Sigma(\overline{n}_{11}, 0) > 0$ . Otherwise,  $\Sigma(\overleftarrow{n}_{11}, 0) = 0$ . Note that when events  $E_1$ ,  $E_{8,1}$ ,  $E_{8,2}$ ,  $E_{10,1}$ , and  $E_{10,2}$  hold simultaneously true, then the event  $S_4$  in (31) holds true, which verifies the statement of Theorem 4. The same procedure can be applied for all the other combinations of interference regimes. This completes the proof.

#### REFERENCES

- [1] V. Quintero, S. M. Perlaza, I. Esnaola, and J.-M. Gorce, "When does channel-output feedback enlarge the capacity region of the two-user linear deterministic interference channel?" in Proc. 11th EAI International Conference on Cognitive Radio Oriented Wireless Networks, Grenoble, France, May. 2016.
- [2] C. E. Shannon, "Two-way communication channels," in Proc. 4th Berkeley Symposium on Mathematical Statistics and Probability, Berkeley, CA, USA, 1961, pp. 611-644.
- [3] R. Ahlswede, "The capacity region of a channel with two senders and two receivers," The Annals of Probability, vol. 2, no. 5, pp. 805-814, Oct 1974
- [4] A. Carleial, "A case where interference does not reduce capacity," IEEE Trans. Inf. Theory, vol. 21, no. 5, pp. 569-570, Sep. 1975.
- [5] H. Sato, "Two-user communication channels," IEEE Trans. Inf. Theory, vol. 23, no. 3, pp. 295-304, 1977.
- A. Carleial, "Interference channels," IEEE Trans. Inf. Theory, vol. 24, [6] no. 1, pp. 60-70, Sep. 1978.
- H. Sato, "The capacity of the Gaussian interference channel under strong interference," IEEE Trans. Inf. Theory, vol. 27, no. 6, pp. 786-788, Nov. 1981.

- [8] T. S. Han and K. Kobayashi, "A new achievable rate region for the interference channel," *IEEE Trans. Inf. Theory*, vol. 27, no. 1, pp. 49– 60, 1981.
- [9] R. H. Etkin, D. N. C. Tse, and W. Hua, "Gaussian interference channel capacity to within one bit," *IEEE Trans. Inf. Theory*, vol. 54, no. 12, pp. 5534–5562, Dec. 2008.
- [10] I. Sason, "On achievable rate regions for the Gaussian interference channel," *IEEE Trans. Inf. Theory*, vol. 50, no. 6, pp. 1345–1356, Jun. 2004.
- [11] G. Kramer, "Outer bounds on the capacity of Gaussian interference channels," *IEEE Trans. Inf. Theory*, vol. 50, no. 3, pp. 581–586, Mar. 2004.
- [12] H.-F. Chong, M. Motani, and H. K. Garg, "A comparison of two achievable rate regions for the interference channel," in *ITA Workshop*, Feb. 2006.
- [13] H.-F. Chong, M. Motani, H. K. Garg, and H. El Gamal, "On the Han-Kobayashi region for the interference channel," *IEEE Trans. Inf. Theory*, vol. 54, no. 7, pp. 3188–3195, Jul 2008.
- [14] C. E. Shannon, "The zero-error capacity of a noisy channel," *IRE Transactions on Information Theory*, vol. 2, no. 3, pp. 8–19, Sep. 1956.
- [15] C. Suh and D. N. C. Tse, "Feedback capacity of the Gaussian interference channel to within 2 bits," *IEEE Trans. Inf. Theory*, vol. 57, no. 5, pp. 2667–2685, May. 2011.
- [16] S.-Q. Le, R. Tandon, M. Motani, and H. V. Poor, "Approximate capacity region for the symmetric Gaussian interference channel with noisy feedback," *IEEE Trans. Inf. Theory*, vol. 61, no. 7, pp. 3737–3762, Jul. 2015.
- [17] V. Quintero, S. M. Perlaza, I. Esnaola, and J.-M. Gorce, "Approximate capacity region of the two-user Gaussian interference channel with noisy channel-output feedback," (Submitted to) *IEEE Trans. Inf. Theory*, Nov 2016. [Online]. Available: https://arxiv.org/pdf/1611.05322.pdf
- [18] A. Sahai, V. Aggarwal, M. Yuksel, and A. Sabharwal, "Capacity of all nine models of channel output feedback for the two-user interference channel," *IEEE Trans. Inf. Theory*, vol. 59, no. 11, pp. 6957–6979, 2013.
- [19] D. Tuninetti, "On interference channel with generalized feedback (IFC-GF)," in *Proc. of International Symposium on Information Theory (ISIT)*, Nice, France, Jun. 2007, pp. 2661–2665.
- [20] S. Yang and D. Tuninetti, "Interference channel with generalized feedback (a.k.a. with source cooperation): Part I: Achievable region," *IEEE Trans. Inf. Theory*, vol. 5, no. 57, pp. 2686–2710, May. 2011.
- [21] D. Tuninetti, "An outer bound region for interference channels with generalized feedback," in *IEEE Information Theory and Applications Workshop (ITA)*, Feb. 2010, pp. 1–5.
- [22] —, "An outer bound for the memoryless two-user interference channel with general cooperation," in *IEEE Information Theory Workshop (ITW)*, Sep. 2012, pp. 217–221.
- [23] A. Vahid, C. Suh, and A. S. Avestimehr, "Interference channels with rate-limited feedback," *IEEE Trans. Inf. Theory*, vol. 58, no. 5, pp. 2788– 2812, May. 2012.
- [24] G. Bresler and D. N. C. Tse, "The two user Gaussian interference channel: A deterministic view," *European Transactions on Telecommunications*, vol. 19, no. 4, pp. 333–354, Apr. 2008.
- [25] C. E. Shannon, "Channels with side information at the transmitter," *IBM Journal of Research and Development*, vol. 2, no. 4, pp. 289–293, Oct. 1958.



Samir M. Perlaza (S'07-M'11-SM'15) is a research scientist with the Institut National de Recherche en Informatique et en Automatique (INRIA), France, and a visiting research scholar at the School of Applied Science at Princeton University (NJ, USA). He received the M.Sc. and Ph.D. degrees from École Nationale Supérieure des Télécommunications (Telecom ParisTech), Paris, France, in 2008 and 2011, respectively. Previously, from 2008 to 2011, he was a Research Engineer at France Télécom - Orange Labs (Paris, France). He has held long-

term academic appointments at the Alcatel-Lucent Chair in Flexible Radio at Supélec (Gif-sur-Yvette, France); at Princeton University (Princeton, NJ) and at the University of Houston (Houston, TX). Dr. Perlaza has been distinguished by the European Commission with an Alban Fellowship in 2006 and a Marie Skłodowska-Curie Fellowship in 2015. He was also one of the recipients of the the Best Student Paper Award at Crowncom in 2009. His research interests lie in the overlap of signal processing, information theory, game theory and wireless communications.



**Iñaki Esnaola** (S'08-M'11) received the M.S degree in Electrical Engineering from University of Navarra, Spain in 2006 and a Ph.D. in Electrical Engineering from University of Delaware, Newark, DE in 2011. He is currently a Lecturer in the Department of Automatic Control and Systems Engineering of The University of Sheffield, and a Visiting Research Collaborator in the Department of Electrical Engineering of Princeton University, Princeton, NJ. In 2010-2011 he was a Research Intern with Bell Laboratories, Alcatel-Lucent, Holmdel, NJ, and in

2011-2013 he was a Postdoctoral Research Associate at Princeton University, Princeton, NJ. His research interests include information theory and communication theory with an emphasis on the application to electricity grid problems.



Jean-Marie Gorce (M'07, SM'14) is Professor at the Université de Lyon, France, and he is also with the Institut National de Recherche en Informatique et en Automatique (INRIA). He received the PhD degree in Electrical Engineering from the Institut national des sciences appliquées (INSA), Lyon, France in 1998. He is a co-founder of the Centre for Innovation in Telecommunications and Integration of Services (CITI) of which he was the director (2009-2014). He was a visiting scholar at Princeton University (NJ, USA) in 2013-2014. He

holds the Internet of Things Industrial and Research Chair of INSA Lyon sponsored by SPIE ICS. He is currently vice delegate for research at INRIA Grenoble, France. He is the scientific coordinator of the experimental facility FIT-CorteXlab. He has been the principal investigator of several French and European sponsored projects related to wireless networks. He is an associate editor of the Eurasip Journal of Wireless Communications and Networking (Springer). His research interests are in wireless networking and communication theory, focusing on realistic modeling, wireless system optimization, and performance assessment considering both infrastructurebased and *ad-hoc* networks. He has co-published more than 150 conference and journal articles.



Victor Quintero received the B.Sc. and the M.Sc degree in electronics and telecommunications engineering from Universidad del Cauca, Popayán, Colombia, in 1999 and 2011, respectively. He is currently on his academic leave granted by Universidad del Cauca to pursuit his PhD studies on electrical engineering at Université de Lyon, Lyon, France. His research interests span information theory, signal processing and wireless communication systems.